THE BARRIERS TO ACCESS FOR MATERNAL HEALTH CARE AMONGST PREGNANT ADOLESCENTS IN THE MITCHELLS PLAIN SUB-DISTRICT

Michelle Olivia Erasmus

A mini-thesis submitted in partial fulfillment of the requirements for the degree of Master in Public Health at the School of Public Health,
University of the Western Cape

Supervisor: Dr Lucia Knight Co-supervisor: Prof Asha George

November 2017

KEYWORDS

Reproductive Health

Adolescent

Teenage

Adolescent Pregnancy,

Access,

Maternal Health Care Services,

Utilization of maternal health care services,

Perceptions,

Health care workers,

Mitchells Plain Sub-district.



ABBREVIATIONS

ART- Antiretroviral therapy

HCW- Healthcare Worker

HIV- Human Immunodeficiency Virus

MTCT- mother-to-child-transmission

MOU- Maternity and Obstetrics Unit

SRH- Sexual and Reproductive Health

UN- United Nations

UNESCO- The United Nations Educational, Scientific and Cultural Organization

UNIVERSITY of the WESTERN CAPE

UNFPA- The United Nations Population Fund

UNICEF-United Nations Children's Fund

WHO- World Health Organization

ii

ABSTRACT

Background and aim: Adolescent pregnancy holds numerous health and social risks for mother and child. Adolescent access to maternal health services is of vital importance to ensure that pregnant adolescents receive timely and effective health care. Evidence suggests that adolescents tend to seek medical care late in their pregnancies and attend fewer than the recommended four antenatal visits. This results in missed opportunities to improve maternal and newborn health due to untreated sexually transmitted diseases and uninhibited mother-to-child HIV transmission, resulting in low birth weight among other poor pregnancy outcomes.

Sub-Saharan Africa has the highest rate of adolescent births worldwide, in keeping with evidence that low-to-middle income countries have the highest rates of adolescent pregnancy. Adolescent pregnancy in South Africa remains a public health concern, and the Mitchells Plain sub-district has a particularly high rate of adolescent pregnancy. The aim of this study was thus to explore the barriers to access for maternal health care services amongst pregnant adolescents in the Mitchells Plain sub-district.

Methodology: This study used an exploratory qualitative design to explore the views of participants. Purposive sampling was used to select participants. In-depth semi-structured interviews were conducted with 10 pregnant adolescents at the Mitchells Plain Maternity and Obstetrics Unit (MOU), as well as 2 nursing staff working at Mitchells Plain MOU. Interviews were recorded and transcribed by the researcher herself. The thematic coding approach was then used to analyse the data.

Results: Results showed that fear of disclosing pregnancy, negative response to pregnancy disclosure, and feelings of shock and disbelief in response to pregnancy served as barriers to access to maternal health care services amongst pregnant adolescents. Negative perceptions of maternal health care services also served as a barrier to access to maternal health care services amongst pregnant adolescents. Negative perceptions included expected mistreatment by nursing staff and expected long waiting times. Many participants also reported poor knowledge of maternal health care services, which was a barrier to access to maternal health care services. Facilitators to access to maternal health care services included support and advice from parents or guardians, as

well as positive interactions with nursing staff. The support from parents and guardians often occurred after an initial negative response to disclosure of the pregnancy.

Conclusion and recommendations: Access to maternal health care services amongst pregnant adolescents is influenced by numerous factors. Societal norms and expectations about adolescent childbearing and knowledge about maternal health care services stand out as key factors in facilitating or hindering access to maternal health care services amongst pregnant adolescents. It is recommended that an inter-sectoral approach be adopted in improving access to maternal health care services amongst pregnant adolescents. This includes incorporating maternal health education into school's life skills curriculum, a well-functioning referral pathway between schools and clinic to facilitate early access to maternal health care services, as well as involvement by the social development sector to facilitate pregnancy disclosure for pregnant adolescents who fear negative response to disclosure of the pregnancy from their parents or guardians.

UNIVERSITY of the

DECLARATION

I, Michelle Olivia Erasmus, declare that *The barriers to access for maternal health care amongst* pregnant adolescents in the Mitchells Plain Sub-district, is my own work, that is has not been submitted before for any degree or examination at any other university, and that all sources I have used or quoted have been acknowledged.

Michelle Olivia Erasmus

Signature. Moreover

Date: November 2017



ACKNOWLEDGEMENTS

Firstly, I give all thanks to God for gracing me with all I needed to complete this thesis. A special thanks to my supervisor, Dr Lucia Knight and co-supervisor Prof Asha George for the valuable guidance, input and motivation throughout this process.

Thank you to the administrative staff at the University of the Western Cape School of Public Health, in particular Corinne Carolissen and Janine Kader.

Thank you to the Department of Health and the staff at Mitchells Plain Maternity and Obstetrics Unit for assisting me through the data collection process. Your helpfulness and welcoming nature made such a big impact on me, and I am so thankful to you.

A heartfelt thank you to all the participants in this study, I am eternally grateful. Thank you for being so willing to open up and speak about your experiences.

Thank you to my parents, Albert and Aloma, for you constant love and support.

To my husband, Marcelino, and my daughters, Madison and Morgan, thank you for your love, support and encouragement.

To my siblings, Desiree, Gradwell and Ricardo, thank you for always cheering me on and encouraging me.

Thank you to my parents-in-law, Marcelle and Bernard Erasmus, as we as well as my sisters-in-law Jamie and Jade Erasmus for your support, and for assisting to look after the children when I needed time to work.

DEDICATION

This work is dedicated to the memory of my departed grandparents, Edward Richards, Cupido Fransman and Sarah Fransman.

This work is also dedicated to my parents, Albert and Aloma Fransman, my grandmother Joan Richards, my husband Marcelino, and my siblings, Desiree, Gradwell and Ricardo. Your love and support are invaluable.

Finally, I want to dedicate this work to my daughters, Madison and Morgan, my niece Zoe, and my nephews Matthew and Kyler. May you be inspired to reach for your dreams.



TABLE OF CONTENTS

Keywords			
Abbreviations			
Abstract			
Declaration			
Acknowledgements			
Dedication			
Table of contents			
Chapter 1 Introduction			
1.1	Background Information	1-4	
1.2	Context	4-5	
1.3	Problem Statement	5-6	
1.4	Rationale	6	
1.5	Outline of ThesisUNIVERSITY of the	7	
Chapter 2 Literature Review			
2.1	Introduction	8	
2.2	2 Risk factors for adolescent pregnancy		
2.3	Potential risks associated with adolescent pregnancy	9-11	
2.4	Perceptions concerning access to health services amongst adolescents		
2.4.1	Inconvenient clinic hours and impact on school attendance	11-12	
2.5	Fear of disclosure of pregnancy	12	
2.6	Relationship with healthcare workers		
2.7	Knowledge regarding sexual and reproductive health		
2.8	8 Decision-making power to attend antenatal clinics amongst adolescents 1		
2.9	9 Summary		
Chapter 3 Methodology			

3.1	Introduction	17
3.2	Aim	17
3.3	Objectives	17
3.4	Study Design	17-18
3.5	Research Setting	18-19
3.6	Study Population	19
3.7	Sampling Procedure	19-20
3.8	Data Collection	20-21
3.9	Data Analysis	21
3.10	Rigour	21-22
3.11	Ethical Considerations	22-23
Cha	pter 4 Findings	24
4.1	Introduction	24
4.2	Sociodemographic profile of participants	25-26
4.3	Fear of disclosing pregnancy	26-28
4.3.1	Negative initial response from parents or guardians to pregnancy	28-30
4.4	Feelings of shock and disbelief in response to pregnancy	31-32
4.5	Perceptions and experiences of maternal health care services	32
4.5.1	Rumours of rude nursing staff	32-33
4.5.2	Interaction with nursing staff	33-36
4.5.3	Expected long waiting times	36-37
4.6	Poor self-reported knowledge of maternal health care services	37-38
4.7	Conclusion	38-39
Chapter 5 Discussion		
5.1	Introduction	40
5.2	Fear of disclosure of pregnancy	40-41

2.2.1 Negative initial response to pregnancy from parents		
5.3 Feelings of shock and disbelief in response to pregnancy	43	
5.4 Perceptions and experiences of maternal health care services	43-44	
5.4.1 Rumours of rude nursing staff		
5.4.2 Interaction with nursing staff	45-46	
5.4.3 Expected long waiting times	46-47	
5.5 Poor self-reported knowledge of maternal health care services		
5.6 Limitations		
5.7 Summary	49	
Chapter 6 Conclusion	50	
6.1 Conclusion	40-51	
6.2 Recommendations	51-52	
References	53-68	
Appendices	69	
Appendix 1: Information Sheet Pregnant Adolescents		
Appendix 2: Information Sheet Parents or Guardians		
Appendix 3: Information Sheet Nursing Staff		
Appendix 4: Informed Consent Form		
Appendix 5: Informed Consent: Parent Permission Form	79	
Appendix 6: Assent Form: Pregnant Adolescents	80	
Appendix 7: Interview Guide: Pregnant Adolescents		
Appendix 8: Interview Guide: Nursing Staff		
<u>List of tables</u>		
Table 1: Age distribution of participants		
Table 2: Relationship status of participants		

CHAPTER 1-INTRODUCTION

1.1 BACKROUND INFORMATION

An adolescent is defined as any person between 10 and 19 years of age (WHO, 2017). Of all annual births worldwide, 11% occur in girls aged 15 to 19 years of age (WHO, 2014). This amounts to approximately 16 million births amongst adolescents. Of these births, 95% occur in low to middle income countries. The highest birth rates amongst adolescents are found in sub-Saharan Africa, with 200 per 1000 births in sub-Saharan Africa occurring amongst the adolescent age group (UNICEF, 2016). Adolescent birth rates in developing countries are more than double the adolescent birth rates in most developed countries, with developed countries such as Japan showing adolescent birth rates of less than 1% annually (WHO, 2008).

Despite efforts to decrease maternal mortality, sub-Saharan Africa still has the highest maternal mortality rates worldwide (UN, 2014). In terms of maternal mortality, maternal disorders are the second leading cause of death amongst adolescent girls, with adolescent girls facing a higher risk of complications and death from pregnancy and childbirth compared to older women (AbouZahr, 2013; WHO, 2014). Annually approximately 15% of maternal deaths worldwide occur in adolescent girls. However, Africa has a higher maternal mortality amongst adolescents in comparison with the global average, with 26% of maternal deaths in Africa occurring amongst adolescents. Furthermore, sub-Saharan Africa also has the highest rate of unsafe abortions worldwide (AbouZahr, 2013). According to AbouZahr (2013), it is estimated that one out of every four unsafe abortions in sub-Saharan Africa occurs in the adolescent population. However, it is thought that many adolescent pregnancies, births and abortions in sub-Saharan Africa may be undocumented, resulting in difficulties in accurate data collection regarding adolescent pregnancies.

Though South Africa saw an initial decline followed by stagnation in adolescent pregnancy rates over recent years, it still has a relatively high adolescent birth rate, with 49 per 1000 births occurring amongst adolescent females in South Africa (World Bank, 2015). The percentage of females between the ages of 15 and 19 years who have started bearing children has remained at

16% in 1998 as well as in 2016 (Department of Health, 2017). Adolescent pregnancy in South Africa is most prevalent amongst the second poorest percentage of the population, and least prevalent amongst the wealthiest percentage of the population (Department of Health, 2017). In 2015, KwaZulu Natal had the highest number of births amongst females between the ages of 15 and 19 years, with 30 233 births recorded (Statistics South Africa, 2016). The Western Cape had had recorded 10 382 births amongst females between 15 and 19 years of age for 2015, the sixth highest rate amongst this age group in South Africa. The lowest number of births amongst females between 15 and 19 years of age for 2015 was found in the Northern Cape, with 4 157 births (Statistics South Africa, 2016).

Mitchells Plain sub-district in Cape Town, has a population of 310 485 (City of Cape Town, 2013). In 2012/2013, Cape Town in the Western Cape had a birth rate of 54.6 per 1000 amongst women under the age of 18 years, with Mitchells Plain sub-district showing a birth rate of 56 per 1000 deliveries for women under 18 years of age, relatively higher than the national average (Western Cape Government Provincial Treasury, 2013). The City of Cape Town had a maternal mortality ratio of 78.9 per 100 000 live births in 2012/2013, but there was very scanty data regarding maternal mortality amongst adolescents. It is suspected that maternal deaths, including amongst adolescents, are under-reported or misclassified on death certificates (Groenewald et al, 2008).

Adolescent pregnancy is a multi-faceted issue, and has been shown to have various negative consequences. Pregnant adolescents are more prone to premature labour, obstetric complications and HIV transmission (UN, 2014; AbouZahr, 2013; WHO, 2014; UNFPA, 2013). Children of adolescent mothers also have worse outcomes than children of older mothers, with issues such as low birth weight and higher mortality rates (WHO, 2014; Althabe et al., 2013).

Adolescent pregnancy in South Africa is indicative of the fact that adolescents are engaging in high risk sexual behaviours, which in turn puts them at risk of contracting sexually transmitted diseases such as HIV (Ehlers et al., 2000; Kanku & Mash, 2010). Pregnant adolescents are thus more likely to be exposed to HIV. In keeping with this, Christofides et al. (2014) report that early adolescent pregnancy is associated with increased incidence of HIV. Furthermore, adolescents have shown slower uptake of antenatal ART, as well as higher mother-to-child transmission of

HIV (Fatti et al., 2014). This makes early access to maternal healthcare services, amongst pregnant adolescents, a priority in the management of HIV.

The social and economic effects of adolescent pregnancy are also far-reaching (UNESCO, 2014). Adolescent pregnancies are more common in poorer communities with high rates of crime, violence, alcohol and drug abuse (Klein, 2005; UNESCO, 2014; Vundule et al., 2001; UNFPA, 2013). Adolescent mothers very often do not finish their schooling, leading to fewer work opportunities or a reliance on low-income work (UNFPA, 2013; UNESCO, 2014; Flanagan et al., 2013). This has an impact on the livelihood of the adolescent mother and child, the family unit, as well as the community as a whole, making it more difficult for individuals and communities to achieve economic growth and move out of poverty.

Given the physical, psychological, social, and economic implications of adolescent pregnancy, it is clear that adolescent pregnancy is a public health problem which requires collaboration between different sectors (UNFPA, 2013). This includes collaboration between health, education and social development sectors. Access to comprehensive care during pregnancy is thus imperative for this vulnerable group. However, adolescents are known to under-utilize sexual and reproductive health (SRH) care services, including antenatal care (Worku & Woldesenbet, 2016; Alli et al., 2012; Burack, 2000). Pregnant adolescents are also known to access maternal health care services very late in pregnancy and show poor attendance at antenatal clinics (Tsawe & Susuman, 2014; Atuyambe et al., 2008). This creates a missed opportunity for HIV counselling, testing and treatment, and prevents early detection of potential problems in preparation for labour. Access to maternal health care services is thus of utmost importance for the health of the adolescent mother and her baby.

The SRH rights of adolescents in South Africa are protected by law, yet adolescents still face numerous barriers to accessing SRH care services (Hoopes et al., 2015). Sexuality amongst adolescents is still a controversial topic in many communities, and adolescent pregnancy is seen as something shameful (James et al., 2012; Ilika & Anthony, 2004), creating an environment in which adolescents fear disclosing their pregnancy. This makes adolescent pregnancy and the provision of antenatal care to adolescents a complex issue for both patient and healthcare provider.

Access to healthcare is defined as "the opportunity to identify healthcare needs, to seek healthcare services, to reach, obtain or use healthcare services and to have healthcare needs fulfilled" (Levesque et al., 2013:8). This means that factors such as the ability to recognise that one has a particular health care need, seek health care, reach healthcare services, pay for health care services and engage with healthcare providers are important dimensions of access. When any of these elements are compromised, it acts as a potential barrier to access to healthcare services.

Keeping the above-mentioned definition of access in mind, I will be exploring the issue of adolescent access to maternal health care services from the perspective of pregnant adolescents as well as healthcare providers in the Mitchells Plain sub-district.

1.2 CONTEXT

In 2016, 16% of South African girls between the ages of 15 and 19 years had started childbearing (Department of Health, 2017). Cape Town in the Western Cape has a high rate of adolescent pregnancy, with a birth rate of 54.6 per 1000 amongst women under the age of 18 years in 2012/2013 (Western Cape Government Provincial Treasury, 2013). The Mitchells Plain subdistrict in Cape Town, however, had a birth rate of 56 per 1000 deliveries for women under 18 years of age for 2012/2013, which raises concern about the rate of adolescent childbearing in Mitchells Plain.

Mitchells Plain sub-district in the Western Cape, Cape Town, has a population of 310 485 (City of Cape Town, 2013). Mitchells Plain sub-district in the Western Cape, Cape Town, has a population of 310 485 (City of Cape Town, 2013). The Mitchells Plain community is predominantly coloured, with 95% of households living in formal dwellings, 99% of dwellings with access to piped water and electricity and 96% of households with flush toilets connected to a public sewer system. Despite this, the Mitchells Plain community has high levels of crime, gangsterism, unemployment and poverty (Department of Provincial and Local Government, 2011). According to the Department of Provincial and Local Government (2011), approximately 48% of households in Mitchells Plain earn below the poverty line, and 43% of residents who fall within employment age ranges are employed. Furthermore, there are high levels of overcrowding in certain areas. Large numbers of residents make use of a well-established public transport

system, in the form of buses, taxis and trains (Department of Provincial and Local Government, 2011).

Mitchells Plain is divided into approximately 9 sub-areas, and each sub-area has a primary health care facility located within 2.5 kilometres (Department of Provincial and Local Government, 2011). It has one private hospital and a district hospital, as well as numerous private general practitioners (GP) and pharmacies. Contraceptives, HIV testing and pregnancy testing is available at primary health care facilities as well as private health care providers and pharmacies. Mitchells Plain MOU services residents from all sub-areas in Mitchells Plain, as well as residents from Khayelitsha, Philippi, Samora Machel and numerous informal settlements that have been erected around the Mitchells Plain area.

The City of Cape Town had a maternal mortality ratio of 78.9 per 100 000 live births in 2012/2013. However, there is very little available data about maternal mortality amongst adolescents in particular. This could be due to possible errors in classification of causes of death amongst pregnant women, including adolescents (Groenewald et al., 2008). In light of the abovementioned challenges, it is evident that reproductive health and access to maternal health services amongst adolescents needs to be explored. TY of the

WESTERN CAPE

The high rate of unplanned adolescent pregnancy in Mitchells Plain is in keeping with literature in that communities with high rates of crime, violence and unemployment also have high rates of adolescent pregnancy (Brahmbatt et al., 2014; Thobejane, 2015).

1.3 PROBLEM STATEMENT

Adolescent pregnancy has many potential negative implications, including higher risk for maternal mortality, serious adverse reproductive health problems, higher risk of mother-to-child transmission of HIV, mental health problems including depression, poorer health outcomes for children born to adolescent mothers and high development costs for communities in perpetuating the cycle of poverty (UNFPA, 2013). Adolescent mothers also face an array of socio-economic difficulties such as pressure to leave school, difficulty finding gainful employment, strained relationships with family and stigma from community (Ehlers et al., 2000). It is thus imperative

that research be conducted into access to services amongst pregnant adolescents in order to improve their health, social, and economic outcomes.

Pregnant adolescents may initiate antenatal care late in pregnancy and are more likely to attend fewer than the recommended four antenatal visits (Atuyambe et al., 2008; Phafoli et al., 2007). This puts this group at higher risk of adverse outcomes due to pregnancy. It is thus important that adolescent mothers receive timely and accessible health care and social support. However, there is little research into factors which influence access to maternal health services amongst adolescents. Mitchells Plain sub-district has a particularly high adolescent pregnancy rate. The aim of this study is thus to explore the barriers regarding access to maternal health services amongst pregnant adolescents in Mitchells Plain sub-district.

1.4 RATIONALE

Mitchells Plain has a high rate of adolescent pregnancy (Western Cape Government Provincial Treasury, 2013). The pregnant adolescent population may have late access to maternal health care services (Phafoli et al., 2007). Poor access and underutilization of maternal health care services is linked to poorer outcomes amongst pregnant adolescents and their infants.

The purpose of this study is to gain a better understanding into the perspectives of adolescents regarding access to maternal health services, as well as to gain understanding into how healthcare workers experience providing care to pregnant adolescents. Once we have a good understanding of how adolescents perceive access to maternal health care, this knowledge can be used to find ways of improving utilization of reproductive health care services for adolescents. This will help reduce maternal mortality and improve health outcomes for the adolescent mother and child. This data could be particularly useful to facility managers at Mitchells Plain MOU, as well as other health care workers (HCW's) to improve reproductive health care services to adolescents.

1.5 OUTLINE OF THESIS

This thesis is arranged as follows:

Chapter 1 introduces the issue of adolescent pregnancy and explains the rationale behind pursuing this research. Chapter 2 contains the literature review, which gives an understanding into what literature is available about adolescent pregnancy and access to SRH services amongst adolescents. Chapter 3 contains details of the research aims, objectives and study design, and provides details about the research methodology of the study. Chapter 4 presents the findings of the study. Chapter 5 follows, which contains the data analysis, discussion, and interpretation of the results. Lastly, chapter 6 presents the conclusion of the study and provides further recommendations based on the findings of the study about improving access to maternal health care services amongst adolescents.



CHAPTER 2- LITERATURE REVIEW

2.1 INTRODUCTION

This literature review focuses on adolescent access and utilization of sexual, reproductive and maternal health care services. Further available literature regarding risks of adolescent pregnancy and perceptions amongst adolescents regarding access to sexual and reproductive health (SRH) care services is also explored. This includes inconvenient clinic hours and fear of disclosure of pregnancy. The relationship between health care workers (HCW's) and adolescents may influence adolescent utilization of health care services, and available literature regarding is therefore be reviewed.

Lastly, I will explore the literature about the knowledge regarding sexual and reproductive health amongst adolescents, as well as the decision-making power of adolescents in seeking out health services.

2.2 RISK FACTORS FOR ADOLESCENT PREGNANCY

Unplanned adolescent pregnancy is an issue that affects many countries across the world. Of all births worldwide, 11% occur amongst adolescents (WHO, 2014). Furthermore, data from the World Health Organisation (WHO) shows that 95% of adolescent births occur in low to middle income countries, with sub-Saharan Africa bearing the highest number of adolescent births. In South Africa, the birth rate amongst adolescents is 49 per 1000 (World Bank, 2015).

UNIVERSITY of the

Studies have shown that poor socio-economic conditions are associated with higher risk for unplanned adolescent pregnancy. Factors such as high levels of crime and violence, high levels of unemployment and poverty are all associated with high rates of unplanned adolescent pregnancy (Brahmbatt et al., 2014). Thobejane (2015) conducted a qualitative study in Limpopo, South Africa to explore factors which may contribute to teenage pregnancy, and identified poverty and family struggles as risk factors to unplanned adolescent pregnancy. Nkani and Bhana (2016) further state that many adolescent mothers come from poorer families, and the financial implications of caring for a child places further stress on household finances. The high

rate of adolescent pregnancy as well as the high rates of poverty and unemployment in Mitchells Plain sub-district is thus in keeping with trends described in the literature.

Understanding the socio-economic circumstances of individuals and communities is vital when addressing the issue of access to health care services (Dookie & Singh, 2012). Poverty and poor socio-economic conditions are often associated with poor access to health services (Copland et al., 2011). Furthermore, the adolescent population often faces difficulties in accessing sexual and reproductive health services, particularly those from poorer socio-economic conditions (Ehlers et al., 2000; Alli et al., 2012). This consequently results in underutilization of SRH care services amongst adolescents (Chandra-Mouli et al., 2014; Ehlers et al., 2000).

High rates of unplanned pregnancies may be testament to the fact that adolescents underutilize SRH care services, amongst other social drivers of adolescent pregnancy such as poverty. An unplanned pregnancy means that the pregnant adolescent engaged in unprotected sex and either did not use or incorrectly used birth control measures. Pregnant adolescents may therefore continue the pattern of underutilization of SRH services during the pregnancy (Burack, 2000). It is thus of utmost importance that reasons for poor access and poor utilization of maternal health care services be further investigated (Chandra-Mouli et al., 2013).

WESTERN CAPE

2.3 POTENTIAL RISKS ASSOCIATED WITH ADOLESCENT PREGNANCY

Adolescent pregnancy poses numerous risks for mother and child (UNFPA, 2013). These include biological, psychological, social and economic risks. Complications that directly stem from pregnancy or childbirth are the second leading cause of death amongst adolescent girls globally, with adolescent girls facing a higher risk of complications and death from pregnancy and childbirth compared to other women (AbouZahr, 2013; WHO, 2014).

Adolescent mothers have a higher risk of going into premature labour, and have a higher risk of having low birth weight babies in comparison to women older than 20 years of age (Fraser et al., 1995; Althabe et al., 2015). This trend may result from poor prenatal care, poor access and poor utilization of maternal health care services, as well as risk behaviour such as smoking during pregnancy (Klein, 2005). Factors such as access to and utilization of maternal health care services, poor nutrition during pregnancy, risk behaviour such as consuming alcohol and

smoking and general poor prenatal care may therefore be underlying factors to premature labour and low birth weight amongst adolescent mothers, rather than maternal age alone (Taffa & Obare, 2004; Raatikainen et al., 2005). This highlights the importance of early access to and utilization of maternal health care amongst pregnant adolescents.

In a randomized control trial in the Eastern Cape exploring the link between HIV infection and adolescent pregnancy, Christofides et al. (2014) found that early adolescent pregnancy was associated with an increased incidence of HIV. Overall, the study found that women who acquired HIV over the two years of follow-up were more likely to report early adolescent pregnancy than adult women and adolescents older than 15 years. Those with early adolescent pregnancy were three times more likely to acquire HIV than adult women. This was due to high risk sexual behaviour, greater number of sexual partners as well as greater age difference between those who reported early adolescent pregnancy and their partners. Dellar et al. (2015) conducted a series of case studies and reviewed the literature to find that early sexual debut and early pregnancy were identified as risk factors for HIV infection among young women in South Africa. Regions where adolescent pregnancies are prevalent show high rates of HIV prevalence, and these two issues therefore appear to be interrelated (Rutenberg et al., 2003).

Research suggests that adolescent mothers have a higher rate of mother-to-child-transmission (MTCT) of HIV than adult women (Horwood et al., 2013; Fatti et al., 2014). In a study by Horwood et al. (2013) in KwaZulu Natal, as well as a study by Fatti et al. (2014) in the Eastern Cape, it was found that the increased rate of MTCT amongst adolescents were found to be related to poor access to maternal health care services and poor adherence with ART regimes, rather than biological factors. Poor access to maternal health care services means that many pregnant adolescents do not get tested for HIV early during the pregnancy, leading to delays in commencing treatment if the pregnant adolescent is HIV positive. Poor access to health services may also lead to poor adherence to ART regimes during pregnancy, which results in increased MTCT rates amongst HIV positive adolescents.

Given the available evidence, we can see that the adolescent population that is at risk for unplanned pregnancy is also at risk of HIV infection as well as MTCT. High rates of unplanned

adolescent pregnancy show that adolescents are engaging in unprotected sexual activity, which heightens the risk of HIV infection (Mchunu et al., 2012).

2.4 PERCEPTIONS CONCERNING ACCESS TO HEALTH SERVICES AMONGST ADOLESCENTS

2.4.1 INCONVENIENT CLINIC HOURS AND IMPACT ON SCHOOL ATTENDANCE

Despite efforts to improve SRH knowledge amongst adolescents, research shows that adolescents still tend to underutilize SRH services (Tegegn et al., 2008). Young people may encounter many difficulties when accessing SRH services, which leads to decreased utilization (Lesedi et al., 2011). In a study in Limpopo, Mushwana et al. (2015) found that pregnant as well as non-pregnant high school girls felt that health care services were not readily available. Respondents indicated that clinics were open at times when they were still at school, making access to clinics difficult. Young people of both genders reported dissatisfaction with opening and closing times of clinics. Newton-Levinson et al. (2016) conducted a systematic literature review in low and middle-income countries regarding barriers to access to health care services for STI's amongst adolescents of both genders, and found that adolescents identified inconvenient clinic hours as a barrier to access to services for treatment of STI's. Alli et al. (2012) had similar findings in a study in KwaZulu Natal, South Africa, highlighting the effect of inconvenient clinic hours on access to SRH services amongst adolescents. Even though the aforementioned studies focus on reproductive health care services in general and not particularly on antenatal care, it highlights the fact that clinic timings are critical for the adolescent population.

Due to opening and closing times of clinics, attending antenatal clinics means that pregnant adolescents will miss school. The rate of school leaving amongst pregnant adolescents is high, and perceived difficulties with balancing antenatal clinic attendance with school attendance may contribute to poor attendance at antenatal clinics amongst adolescents, especially in early stages of pregnancy (Pell et al., 2013; UNESCO, 2014). Adolescent mothers who miss class because of medical appointments or other issues related to pregnancy or motherhood do not usually receive

assistance from teachers at school to catch up with classwork, and there are no programs in South African schools to assist adolescent mothers (Chigona & Chetty, 2007).

Apart from opening and closing times, adolescents are also unhappy about spending long hours waiting for a consultation at clinics (Mayeye et al., 2010; Godia et al., 2014; Onokerhoraye & Dudu, 2017). Lengthy waiting times creates dissatisfaction with services and adds to adolescents' reluctance to visit health care facilities for sexual and reproductive health care needs.

2.5 FEAR OF DISCLOSURE OF PREGNANCY

In the South African context, adolescent pregnancy is often viewed as undesirable, and is often a stressful time for adolescents and their caregivers (Richter & Mlambo, 2005). In a study in Mafikeng, South Africa, Mwaba (2000) found that adolescents of both genders viewed adolescent pregnancy as a negative event, which tarnishes the reputation of the family. An expected negative reaction by family and the community may mean that adolescents conceal their pregnancy for as long as possible (Hill et al., 2015; Phafoli et al., 2007). In many cases, pregnant adolescents encounter violence, anger and breakdown in relationships with parents or caregivers (Hill et al., 2015; Phafoli et al., 2007; Ilika & Anthony, 2004). The delay in disclosure of pregnancy leads to a significant delay in seeking antenatal care.

In a qualitative study in Limpopo, Richter and Mlambo (2005) found that adolescents were reluctant to visit clinics for reproductive health services due to fear of being recognized by community members, who may then inform their parents or caregivers that they have accessed reproductive health care. Adolescents therefore forego accessing SRH services in order to conceal any sexual activity (Rukundo et al., 2015; Ekong, 2016; Onokerhoraye & Dudu, 2017; Woog et al., 2015).

Despite the fact that sexual activity amongst adolescents is not socially acceptable in many communities, many adolescents are still sexually active (Gevers et al., 2013). This creates a situation where adolescents feel compelled to hide their sexual experiences from parents and older community members. The pregnant adolescent may therefore ignore, deny or dismiss early signs of pregnancy, leading to late confirmation of pregnancy and late consultation for antenatal care (Rukundo et al., 2015).

2.6 RELATIONSHIP WITH HEALTH CARE WORKERS

The relationship between adolescents and HCW's regarding sexual, reproductive and maternal health services may be a contributing factor to access and utilization of these services amongst adolescents. In a cross-sectional, descriptive study in the Eastern Cape, South Africa, Mayeye et al. (2010) found that adolescent respondents of both genders were unhappy with staff attitudes towards them. Wood and Jewkes (2006) had similar results in Limpopo, with adolescent girls reporting that nurses spoke to them in a disrespectful manner, resulting in adolescents avoiding attending clinics for their SRH care needs. Fear of visiting clinics due to an anticipated negative reception by clinic staff as well as older women at clinics serves as a major barrier to utilization of reproductive health services amongst adolescent girls (Erulka et al., 2005; Mfono, 1998; Durojaye, 2009). Having specific adolescent clinics does not guarantee that adolescent perceptions of HCW's will be positive, as adolescents who attend accredited adolescent-friendly clinics still report feeling disrespected and judged by HCW's when seeking sexual and reproductive health care services (Mathews et al., 2015).

Healthcare workers also face difficulties when providing sexual and reproductive health services to adolescents. Service providers to adolescents have their own personal beliefs about sexual activity amongst adolescents, and this may influence the manner in which they engage with adolescents (Alli et al., 2012). When HCW's feel that adolescents should not be engaging in sexual activity, they may adopt a harsh approach towards adolescents as a way to discourage adolescents from engaging in sexual activity (Godia et al., 2013; Alli et al., 2012). Even though law in South Africa allows adolescents to freely access sexual and reproductive healthcare services, HCW's may act as gatekeepers to these services amongst adolescents based on religious and moral belief systems (Morwe et al., 2014). In situations where HCW's express their disapproval of the adolescent clients' pregnancy, it creates a divide between the HCW and adolescent (Wiemann et al., 2005). In other cases, HCW's are aware of the fact that pregnant adolescents face numerous challenges but feel that they do not possess the necessary expertise to address the needs of pregnant adolescents (Rukundo et al., 2015; Kibombo et al., 2008).

A cross-sectional study in Botswana by Lesedi et al. (2011) revealed that a positive relationship between adolescents and healthcare providers is often associated with increased utilization of

SRH care services. Furthermore, adolescents prefer to receive advice regarding SRH from service providers whom they deem to be relatable and up to date with current norms (Kibombo et al., 2008). The health care system thus has a potentially important role to play not only in providing health care services, but also in providing counselling, support and changing social norms with regards to the management of adolescent pregnancy.

2.7 KNOWLEDGE REGARDING SEXUAL AND REPRODUCTIVE HEALTH

In women of all ages, good knowledge of maternal health care services has been linked with increased utilization of maternal health care services (Nisar et al., 2003; Taffa & Obare, 2004). Higher level of education is also positively associated with increased utilization of antenatal services. Many pregnant adolescents, however, have lower levels of education, either dropping out of school due to pregnancy or leaving school prior to pregnancy (Maness et al., 2016). Many pregnant adolescents have also been found to have poor knowledge regarding SRH, including antenatal care (UNESCO, 2014; Phafoli et al., 2007; Tsawe & Susuman, 2014). This lack of knowledge is thought to contribute to late access to antenatal care amongst adolescents. Poor insight into the physical and emotional changes that they are experiencing during pregnancy, as well as poor knowledge regarding reproductive health in general, may also lead pregnant adolescents to inadvertently engage in harmful practices during pregnancy, such as smoking and unhealthy nutritional practices (Coinco, 2010; Richter & Mlambo, 2005).

Inaccurate information received concerning SRH may also lead to unintended pregnancy (Henry & Fayorsey, 2002). In a study in Cape Town, Vundule et al. (2001) found that receiving information regarding SRH appeared to assist in preventing unplanned adolescent pregnancies. Receiving information regarding SRH from a medical professional was also linked to increased use of these services. However, receiving sexual health information from schools seems to be associated with higher risk for adolescent pregnancy, leading to doubts with regards to the detail and quality of information given by educators (Henry & Fayorsey, 2002; Vundule et al., 2001). Though these studies are not specific to utilization of antenatal services amongst adolescents, it shows that level of knowledge as well as source of knowledge and quality of information may influence patterns of utilization of reproductive health care services amongst adolescents.

Knowledge regarding SRH on its own, however, may not be a strong enough pull factor towards utilization of reproductive health care services amongst adolescents (Thobejane, 2015). The complex nature of adolescent sexuality means that one factor alone does not predict utilization of SRH services, but accurate information regarding SRH is a crucial component in the decision to seek out these services (Haider, 2008; Wood & Jewkes, 2005).

2.8 DECISION-MAKING POWER TO ATTEND ANTENATAL CLINICS AMONGST ADOLESCENTS

Adolescent girls are commonly in a disadvantaged position in terms of decision-making power to attend antenatal services as well as use of contraception before becoming pregnant (UNPFA, 2013; Lenters et al., 2015). Even though the pregnant adolescent may wish to attend antenatal clinics early, the fact that family, friends and the community may react negatively to pregnancy means that pregnant adolescents are disempowered as a result of stigma (Lenters et al., 2015; Haider, 2008). Adding to this is the fact that adolescents typically are still in many cases financially reliant on parents, guardians or partners/spouses, which means that an expectation is created for the adolescent to conform to the rules and standards set by those who provide food, shelter and sustenance for them (Lenters et al., 2015). Many pregnant adolescents access maternal health care services only after pregnancy has been disclosed to parents, guardians or partners (Phafoli et al., 2007) which demonstrates the extent to which the decision-making power of pregnant adolescents become limited by societal expectations. It is widely thought that parents should monitor adolescents closely in order to ensure the best possible reproductive outcomes for adolescents, but an autocratic parental approach to sexual health may in fact serve as a barrier to access to sexual and reproductive care services amongst adolescents (Macleod &Tracey, 2010). Furthermore, fear of deviating from religious and societal norms plays a significant role in shaping health-seeking behaviour amongst adolescent girls (AbouZahr, 2013).

Research suggests that parents with lower levels of education are more reluctant to discuss SRH matters with their children (Nyarko et al, 2014). Low levels of parental education means that many pregnant adolescents are already in a socio-economically disadvantaged position with limited resources, and being unable to communicate honestly about pregnancy and other matters regarding sexual health further disempowers pregnant adolescents (Phafoli et al., 2007).

2.9 SUMMARY

Adolescent access to maternal health care services is a complex issue that involves many factors. Poverty has been identified as an important driver of both adolescent pregnancy and poor access to health care services. Lack of knowledge of maternal health care services is a barrier to access to reproductive and maternal health care services amongst adolescents. Due to unease with discussing sexual matters with elders, adolescents are fearful of disclosing of pregnancy to parents and partners. This causes a delay in disclosing their pregnancy, and a delay in seeking antenatal care. A negative perception of HCW's by adolescents in response to pregnancy is also a recurring theme in the literature. Adolescents feel that they are treated harshly by HCW's, and HCW's face difficulty reconciling their personal beliefs regarding adolescent sexual activity with professional responsibilities.

Inconvenient clinic hours are cited as a barrier to access to SRH care services amongst adolescents. Clinics are open at times when school is in progress, which means that school-going adolescents need to miss class time, with no support structures at school to assist them to catch up on missed class time. Lengthy waiting times were identified as a barrier to access to SRH care services. The decision-making power of adolescents in seeking out SRH care services is greatly limited due to societal pressures and expectations. Often, adolescents only attend antenatal clinics after disclosure of pregnancy to a parent, guardian or partner, which shows that the decision to seek antenatal care is not made by the pregnant adolescent alone.

CHAPTER 3- METHODOLOGY

3.1 INTRODUCTION

The section that follows focuses on the methodology used for this study. The aims and objectives, study design, research setting, study population, sampling procedure, data collection, data analysis, ethical considerations, and limitations of the study are presented.

3.2 AIM

To explore the barriers to access for maternal health care amongst pregnant adolescents in the Mitchells Plain sub-district.

3.3 OBJECTIVES

The study objectives are as follows:

- •To explore experiences of maternal health services and reproductive health amongst pregnant adolescents

 UNIVERSITY of the
- •To explore perceptions of healthcare providers regarding provision of antenatal care to pregnant adolescents
- •To explore the perceptions amongst pregnant adolescents regarding factors which inhibit and facilitate access to maternal health care services.

3.4 STUDY DESIGN

The research approach I used to meet the aims of objectives of this study was a qualitative approach. This study looks at the issue of access to maternal health services from the perspective of pregnant adolescents, and explores perceptions, attitudes, and beliefs of pregnant adolescents regarding access to health care, which lends itself to a qualitative approach. I wanted to gauge how access to maternal health care services is perceived and experienced by the pregnant adolescent community (Patton & Cochran, 2002), as well as how the beliefs and attitudes towards maternal health services influences access to maternal health care services, and a

qualitative design allowed me to unpack these issues. Furthermore, in order understand the perspective of participants; the research should be context-specific, which a qualitative research design allows for (Draper, 2004 Jack 2006). There is limited research and knowledge concerning adolescent access to maternal health care services in South Africa, thus an exploratory qualitative design is best suited to meet the aims and objectives of this study (Manerikar & Manerikar, 2014).

I used semi-structured interviews, which allowed the participant to freely relay their experiences regarding access to maternal health care services to me, without me predetermining what the participants responses should be (Choy, 2014; Robson 2011). Using this open-ended, flexible technique, I was able to develop an understanding of the perspective of pregnant adolescents at Mitchells Plain Mitchells Plain Maternity and Obstetrics Unit (MOU) regarding how they experience and perceive access to maternal health care services. It also allowed nursing staff to recount their experiences of providing care to pregnant adolescents.

Data collection took place from 16 January 2017 until 20 January 2017. Interviews took place immediately after participants were recruited. The study was finalized in November 2017.

3.5 RESEARCH SETTING

UNIVERSITY of the WESTERN CAPE

This study was conducted at Mitchells Plain Maternity and Obstetrics Unit (MOU). The MOU is part of the Mitchells Plain Community Health Centre. Mitchells Plain was established in the 1970's during the apartheid regime as part of mass evictions and relocation of non-whites (Department of Local Government, 2011). The community was initially built to house a population of 250 000, but the population has since grown to 310 485 people (City of Cape Town, 2013). However, some literature reports the population of Mitchells Plain to be around approximately 700 000 (Haque, 2002) largely due to the establishment of informal dwellings. This means that the community healthcare centre and MOU has seen a dramatic increase in patient numbers over the years, with no corresponding increase in resources. This has created overcrowding, long waiting times and frustration for patients and staff alike.

The MOU is well situated close to the bus, train and taxi station, therefore increasing convenience for residents who use public transportation. Mitchells Plain MOU offers both

antenatal and postnatal care, including labour and delivery services. The MOU has an adolescent antenatal clinic every Tuesday. However, due to high patient numbers and lack of space at the facility, adolescents are seated in the same area as older women at the MOU, and women of all ages are accommodated on Tuesdays as a result of the high demand for services. Mitchells Plain District Hospital is the closest referral point for the MOU.

3.6 STUDY POPULATION

The participants I have chosen possessed specific characteristics that gave me the most information about the topic (Ritchie et al., 2003), namely that they were pregnant, fell within the adolescent age group and utilized Mitchells Plain MOU. The sample population consisted of pregnant adolescents attending Mitchells Plain MOU, as they were be able to give the most valuable information about which factors facilitated or inhibited their access to maternal health care. I would also made use of key informants, in this case nursing staff providing services within the facility to pregnant adolescents. Key informants are those with specific knowledge about the study population and topic at hand (USAID, 1996). In this case, the nursing staff gave me particular insight into access to maternal health care services by pregnant adolescents. The implications of my sampling strategy are further discussed in the limitations section.

WESTERN CAPE

The reason for the sample size is that this was a small-scale study for the purpose of a minithesis, therefore logistical and financial constraints necessitated a small sample size. The aim of this study is not to generalize, nor be representative as it is qualitative, which means a large sample was not needed.

3.7 SAMPLING PROCEDURE

For this study, I used purposive sampling to find my sample. This means that participants were chosen based on specific characteristics, which would enable me to gain as much information as possible concerning adolescent access to maternal health care services (Ritchie et al., 2003).

Inclusion criteria were pregnant females between the ages of 16 and 19 years, who used Mitchells Plain MOU services, and understood English or Afrikaans. All those below the age of 18 needed to have a parent/guardian present in order to give consent.

Exclusion criteria were pregnant women over the age of 19 years and those between 10-15 years, and anyone who did not understand English or Afrikaans. Though excluding anyone under the 16 years limited my sample, sexual intercourse is legal from the age of 16 and participation is thus less likely to cause undue stress for participants from the age of 16. Pregnant adolescents who were 16 and 17 years of age who did not have a parent or guardian present with them at the clinic at the time of the interviews were also excluded from the study.

The operational manager of the MOU briefed the nursing staff at the MOU about the research project. The nursing staff at the antenatal clinic then assisted to identify potential participants, and notified me when they were about to assess a potential participant, or when there was a potential participant in the waiting area. I was then able to approach the potential participants, explain the study, and commence the informed consent process.

3.8 DATA COLLECTION

To describe the experiences of pregnant adolescents, I conducted semi-structured interviews with 10 pregnant adolescents attending Mitchells Plain MOU. I also interviewed two nurses with at least 2 years' experience of work at Mitchells Plain MOU. This means that 12 participants were interviewed. The interviews with adolescents covered how they experience the maternal healthcare services at Mitchells Plain MOU, as well as identifying and exploring barriers and facilitators to access to maternal healthcare services amongst pregnant adolescents. The interviews with nursing staff explored how nursing staff experience providing healthcare services to pregnant adolescents and what healthcare workers feel the barriers and facilitators to access to maternal healthcare services are amongst pregnant adolescents.

Nursing staff at the MOU identified potential participants, and I then approached the participants as identified by nursing staff. A thorough informed consent process was followed with all participants. For participants who were 16 and 17 years of age, I gained parental consent, and assent from the participant. Once consent was obtained, I conducted interviews with participants at the MOU. Interviews were audio-recorded with the consent of participants. Two nursing staff members volunteered to be interviewed, and a thorough informed consent process was followed with the nursing staff as well.

A semi-structured interview guide was used to conduct the interviews. Semi-structured interviews allowed me to guide the interview with the aim and objectives of the study in mind, but also allowed the participants to express themselves freely without me predetermining their responses (Choy, 2014). I also took field notes throughout the data collection process.

3.9 DATA ANALYSIS

I personally transcribed the audio-recorded interviews, thus all the data was in textual format (Pope et al., 2000). All interviews took place in English, though certain participants used certain Afrikaans and slang phrases. These were translated to English by the researcher for analysis purposes.

The data was analysed manually, using the thematic coding approach. The thematic coding approach entails assigning codes to phrases, expressions or behaviours that have emerged from the data (Robson, 2011). Themes are then created from the codes by grouping together similar codes that have emerged from the data. The thematic coding data analysis approach allows themes to be modified as the data analysis process unfolds.

Using Robson's (2011) key steps in the thematic coding approach, I first became familiar with the data, which included transcribing the interviews and reading the transcripts numerous times. I then created initial codes for segments that were of particular interest. From the initial codes, I then proceeded to create themes from the initial codes. After creating themes, I read the themes multiple times to ensure that it was an accurate reflection of the data. Once I had created themes, common themes were grouped together to create thematic networks. These thematic networks mean that themes were linked together under a main theme or heading. The data was then interpreted to describe the emerging findings.

3.10 RIGOUR

Qualitative research should be an accurate reflection of the participants' experiences and thoughts, and should therefore be carried out in a systematic manner (Robson, 2011; Srivastava & Thomson, 2009). Rigour ensures that the research is carried out systematically and that the research is trustworthy and is true reflection of the participants' experiences.

To ensure rigour, I used triangulation of data sources by interviewing pregnant adolescents who utilize Mitchells Plain MOU as well as interviewing nursing staff who work at Mitchells Plain MOU. This assisted to systematically find common themes from more than one source, increasing credibility of data (Creswell & Miller, 2000).

Personal reflections on the process were kept in a diary, and these reflections were referred to whilst analysing data and writing up the discussion to gauge how the researchers' own experiences might have impacted how data was perceived and interpreted.

I kept track of all research activities and decisions in a research journal to provide an audit trail. I also provided a detailed description of the research setting, study participants, and themes that arose (Creswell & Miller, 2000). The process of the study as well as my findings and interpretation of data was checked and discussed with my supervisor for peer debriefing to minimise bias.

3.11 ETHICAL CONSIDERATIONS

Ethical clearance for this research was obtained from the University of the Western Cape Research Ethics Committee. Once ethical clearance for this project was obtained from the University of the Western Cape, I applied for permission from the Western Cape Provincial Health Research Committee to conduct the research at Mitchells Plain MOU. This included submission of ethical clearance letter from the University of the Western Cape.

The research project was approved by The Western Cape Provincial Health Research Committee. The Western Cape Provincial Health Research Committee forwarded the approval letter to the person in charge of Mitchells Plain MOU. Once approval was obtained, I contacted the person in charge of Mitchells Plain MOU to further discuss the research project, as well as to clarify best suitable dates for data collection to take place.

Before conducting the interviews, a thorough informed consent process was undertaken with each participant. This ensured that they fully understood the nature and purpose of the research, and that their participation in the study is voluntary. All the information required by them was included in the participant information sheet, and each participant was given a participant

information sheet to keep. It was made clear that they are free to withdraw from the study at any stage, without any negative consequences.

During the interview process, I asked the participant a series of questions in order to gain as much information from the participant as possible. This probing and questioning could have potentially caused feelings of anxiety and distress in the participant (Richards & Schwartz, 2002). However, I ensured that each participant was aware that all effort would be made to ensure their comfort during the interview, and that any feelings of distress or anxiety would be dealt with in a dignified manner and that the participants' health and safety would be held in the highest regard.

For those under the age of 18, consent was obtained from their parent/guardian, and the participant gave assent. Their well-being and wishes were respected at all times. Therefore, in cases where the parent or guardian consents to participation, but the participant did not wish to participate, their refusal to participate was accepted and respected.

Due to the sensitive nature of the topic at hand, confidentiality was held in utmost consideration. The autonomy and dignity of all participants was respected at all times. Their anonymity was protected, and no unauthorized access to the audio recordings, transcriptions of interviews and consent forms was allowed. In the event that I became aware of any other further support required by the participant, and in cases where participants requested further help or support, I was prepared to refer them to the relevant person at the MOU who was able to assist them. However, none of the participants required any additional help or support from me. Participation in the study or refusal to participate in the study had no negative effect in the treatment the participants received at the MOU.

CHAPTER 4 FINDINGS

4.1 INTRODUCTION

In this chapter, the findings of the interviews conducted with pregnant adolescents as well as two nursing staff at Mitchells Plain Maternity and Obstetrics Unit (MOU) are presented. The socio-demographic profile of participants is presented, followed by the main themes that emerged from the data.

The first primary theme is fear of disclosing the pregnancy, with the sub-theme negative initial response from parents and guardians to the pregnancy. Fear of pregnancy disclosure and negative initial response has been shown to be a barrier to accessing maternal health care services. The second primary theme is feelings of shock and disbelief in response to finding out about the pregnancy, which also inhibited pregnant adolescents from accessing maternal health care services. The third primary theme presented is perceptions and experiences of maternal health care services, with the sub-themes of rumours of rude nursing staff, interaction with nursing staff and expected long waiting times. Results from this study showed that pregnant adolescents have a negative perception of maternal health care services, and expected to encounter rude nursing staff. This acted as an inhibiting factor to access to maternal health care services, mainly because pregnant adolescent were reluctant to come to the MOU because of negative perceptions of the MOU. However, the experiences of many adolescents at the MOU were not the same as the perceptions and expectations they had had in terms of interactions with nursing staff. Many adolescents reported a positive experience when interacting with nursing staff, but still reported dissatisfaction with waiting times, which was an inhibiting factor to access to maternal health care services. The fourth primary theme is poor self-reported knowledge regarding maternal health care, which was a barrier to access to maternal health care services amongst pregnant adolescents.

4.2 SOCIODEMOGRAPHIC PROFILE OF PARTICIPANTS

The following table shows the age distribution of participants:

Age	Number of participants
16 years	3
17 years	1
18 years	1
19 years	5

Table 1: Age distribution of participants

The relationship status of participants is represented as follows:

Relationship status WESTER	Number of participants N CAPE
Married	2
Engaged	1
In a relationship with boyfriend	4
Unmarried but relationship status not specified	3

Table 2: Relationship status of participants

All those who were married or engaged, were married or engaged to the father of their baby after learning that they are pregnant.

Concerning level of education, five participants left school before completing matric, two participants completed matric and three participants were still attending school.

Of the participants who left school before completing matric, three reported that they stopped attending school before becoming pregnant, and one reported that she left school after learning that she was pregnant. Only one participant reported being employed at the time of the interview.

Five participants reported that they live with their mother; two reported that they live with their grandmother; one reported that she lives with her husband; one reported that she lives with her fiancé and one reported that she lives with her boyfriend and his parents. Only one participant indicated that she lived with her parents and siblings in the household.

Only one participant reported that she had planned her pregnancy after her boyfriend expressed his wish to have children, and the remaining nine participants viewed their pregnancy as unplanned. Of the nine unplanned pregnancies, seven participants were very positive about the pregnancy, describing feelings of excitement and joy, and two participants were still coming to terms with an unplanned pregnancy and did not indicate if the pregnancy was wanted. However, both these participants expressed that they wanted to be sure that the baby was healthy.

4.3 FEAR OF DISCLOSING PREGNANCY CAPE

A common theme amongst pregnant adolescents in this study was fear of disclosing their pregnancy to those around them. Pregnant adolescents were fearful of the reactions of their parents or guardians as pregnancy in adolescence is seen as morally and socially unacceptable in many households and communities (Richter & Mlambo, 2005; Mwaba, 2000). Pregnant adolescents therefore delayed informing their parents or guardians about the pregnancy.

[I felt] sad... and guilty [when I found about the pregnancy] ...because I am still young. P3, 16 years old

Pregnant adolescents also feared judgment from others in the community and older women at the MOU.

[I felt nervous about coming to the MOU], nervous because everybody is like looking at me [because] I am so young and stuff. [I feel] nervous [and] uncomfortable also...

[because] the young ladies sitting around me and...older people [are looking at me]. P4, 17 years old

As a result of trying to hide their pregnancy status, pregnant adolescents therefore delayed coming to the MOU for antenatal care. At the first visit to the MOU, pregnant women open a folder, and the expected due date is confirmed. They then know when to come to the MOU for delivery, and the first visit to the MOU is therefore commonly referred to as "booking".

Um...like, at first I was scared to tell them [my parents] ...about my pregnancy. Then I did not want to come book [come to the MOU for initial visit] or anything. I wanted actually my mommy to come with me, like now. Just...I was scared to tell them that is why I did not book before the time. P4, 17 years old

Going to the MOU before disclosing the pregnancy to parents or guardians means that there is a risk that others may recognize the pregnant adolescent in the community, who may then inform the parents or guardians of the pregnant adolescents' attendance at the MOU. Due to fear of pregnancy being revealed to parents by others in the community, participants delayed coming to the MOU.

I was, um, still afraid [to come to the MOU], just to, someone walking past me that knows me, [is going to] tell my parents [that I am pregnant], so... [I am] actually hiding a bit. P8, 19 years old

Nursing staff at the MOU identified fear of disclosing the pregnancy, as a significant reason for delayed access to maternal health care services, amongst pregnant adolescents

Some of them [pregnant adolescents], they come to book [for delivery] late, just because they were hiding the pregnancy. But if you ask them then they will tell you that I did not want my mum to know, my mum didn't know. P10, nursing sister, 57 years old

Fear of disclosing the pregnancy to parents also meant that participants were unable to ask parents or guardians for transportation or money for transportation to the MOU, as asking to be taken to the MOU or for transport money to the MOU would lead to further questioning by parents or guardians. Participants who are still attending school would also have to explain why

they have missed a day of school if they plan to attend the MOU. Fear of disclosing their pregnancy to parents or guardians thus means that pregnant adolescents are unable to access transport and are unable to negotiate time off school to attend MOU appointments.

[I didn't come to the MOU before my mother knew about my pregnancy] Because...I wouldn't have a taxi fare and how would I explain if I don't go to school? P7, 18 years old

Nursing staff interviewed also revealed that many pregnant adolescents are unable to access the MOU because of fear of disclosing their pregnancy at school as well as missing a day at school.

Some of them [pregnant adolescents] they are still at school, they miss their appointment and then when you ask why didn't you [come to the MOU]? [They say] "I was at school", [When you ask pregnant adolescents] But why didn't you ask [for time off school to come to the MOU]? [They say] "I was scared". P10, nursing sister, 57 years old

Many adolescents are financially dependent on parents or guardians. Parents and teachers also often monitor school attendance. This means that pregnant adolescents are often unable to access the MOU independently without the help or knowledge of parents and guardians. Pregnant adolescents therefore delay seeking antenatal care due to fear of pregnancy disclosure.

4.3.1 NEGATIVE INITIAL RESPONSE FROM PARENTS OR GUARDIANS TO PREGNANCY

Fear of disclosing the pregnancy was also related to uncertainty about the response from parents or guardians to the pregnancy. One participant who had not disclosed the pregnancy to her parents yet and attended the MOU without her parent's knowledge, gave the following response when discussing why she had not disclosed the pregnancy to her parents yet:

My mother, she's a understanding person but my father, he's more, he's the strict one, so I don't know[what] his reaction [to the pregnancy will be], basically. P8, 19 years old

This alludes to the fact that adolescents may anticipate negative responses to the pregnancy from parents, and therefore delay disclosing the pregnancy.

I was scared to tell my mommy and daddy about [the pregnancy] ...and when I told my, my mommy saw that I got fat and that, so I told her [that I am pregnant] and then she got kwaad [angry], and afterwards she was happy. And my daddy didn't talk to me... at all. P3, 16 years old

From this, we can see that some adolescents wait until there are overt signs of pregnancy before disclosing pregnancy, and that adolescents are fearful of negative reactions from parents. These fears are often justified, as many respondents reported that their parents or guardians indeed initially responded negatively when learning about the pregnancy.

At first my mommy scold[ed] at me but then she overcame it, but...my daddy is still cross...a bit. P4, 17 years old

This negative response was often anticipated by the pregnant adolescent and contributed to their fear of disclosing their pregnancy. Participants were afraid that their parents or guardians would be angry or disappointed in them for becoming pregnant. Fear of the reaction of parents or guardians was a powerful deterrent to disclosing the pregnancy, and this caused them to delay disclosing the pregnancy and seeking antenatal care. One participant described feeling fearful of disclosing her pregnancy to her mother because of seeing negative responses of other mothers in her community to the pregnancy disclosure of their own daughters:

Maybe [other pregnant girls] mothers' abuse [them], other [mothers of pregnant girls say to them] ok, you are pregnant, and [still young, so] take all your clothes [and] go. Maybe you going to go where? Which place [are] you going to go? [The] only [place you have to live] is your home. Just imagine going outside, your, your, clothes is not in the suitcase, the way it was in the suitcase, it's just in a plastic bag, plastic bag outside. P12, 16 years old

Witnessing or hearing stories about negative reactions to pregnancy caused pregnant adolescents to expect physical or verbal abuse, and even expulsion from home in response to pregnancy disclosure, making them hesitant to inform parents or guardians about their pregnancy, and also hesitant to seek healthcare and risk disclosing their pregnancy.

Despite an initial negative response to the pregnancy in some cases from parents and guardians, this was often followed by support and advice given to pregnant adolescents to attend antenatal clinics as soon as possible. Most participants indicated that the negative response to the pregnancy was followed by support and advice, though one participant had not disclosed the pregnancy yet, and another participant indicated that her mother advised her to go the MOU, but did not indicate if her mother was supportive or accepting of the pregnancy after the pregnancy. The support and advice was frequently reported to come from mothers and grandmothers.

My ma [grandmother] was disappointed [about the pregnancy] but...she said that...she...can't do anything about it and that she will stand by me. P2, 16 years old

Participants tended to attend the MOU after they had disclosed their pregnancy to parents or guardians, and were most often advised by their mothers or grandmothers visit the MOU as soon as possible.

My mother told me that I should come [to the MOU] because...you need vitamins, etc. P5, 19 years old

Some participants expressed that even though they may have received advice from other sources such as friends regarding antenatal clinic attendance, they really only heeded the advice of mothers and grandmothers regarding antenatal clinic attendance.

I [knew I had to come to the MOU] ...but I like, didn't really take it seriously until my grandmother talked. P9, 19 years old

A feared or experienced negative initial response from parents or guardians thus contributed to delays in pregnancy disclosure and in antenatal clinic attendance, but pregnant adolescents were often encouraged by parents, guardians and partners to attend the MOU after their pregnancy was disclosed. It was noted that some participants appeared to be particularly concerned about how their father would react to the pregnancy, and that fathers tended to remain angry for longer than the mothers of pregnant adolescents did after disclosure of the pregnancy.

4.4 FEELINGS OF SHOCK AND DISBELIEF IN RESPONSE TO PREGNANCY

Another common theme was feelings of shock and disbelief when discovering that they were pregnant. The majority of participants viewed this as an unplanned event and did not expect to become pregnant.

It is very shocking actually cos...I didn't even know what to do, that [is why] I didn't even come early for the...to the clinic for some test, so ya [yes]. It was shocking P7, 18 years old

Feelings of shock and disbelief meant that many participants were uncertain about how to go about seeking assistance to cope with pregnancy, and did not know how to initiate seeking healthcare. Along with shock and disbelief, some participants described feelings of denial.

Actually, I didn't want to come [to the MOU], you see...so they tell me, my fiancé, he did tell me I must come so that we can find out, you see, before it's too late. So I told him ok then... Because I was scared, I don't want to know [about my pregnancy] actually. P6, 19 years old

For many pregnant adolescents, delaying seeking healthcare is a means of coping with the shock of an unplanned pregnancy and avoiding the reality of being pregnant.

UNIVERSITY of the

[Some pregnant adolescents avoid coming to the MOU because they think] "I don't want this baby so if I don't book, I can still think it's not real, you know, it's not there, you know, I don't have to deal with it, it won't be quite as in my face as when I go [to the MOU] and the sister in the antenatal clinic confirms that I'm pregnant and that I have a baby" P11, Nurse, 52 years old.

Some nursing staff, however, feel that many pregnant young people are not necessarily shocked at the pregnancy, but that many adolescents plan the pregnancy in order to access the child support grant offered by government.

I think maybe they do plan this, these children, I think they plan, maybe, the other thing that maybe can contribute is this SASSA thingy, because they must get the money, also for the grant...maybe... P10, nursing sister, 57 years old

None of the pregnant adolescents in this study mentioned access to social grants and only one participant in this study indicated that her pregnancy was planned. It appears that for the majority of participants, feelings of shock and disbelief lead to difficulties in making decisions about seeking antenatal care early in pregnancy, leading to delays in accessing maternal health care services.

4.5 PERCEPTIONS AND EXPERIENCES OF MATERNAL HEALTH CARE SERVICES

Participants in this study generally had negative perceptions of the MOU. These perceptions were generated by word of mouth in the community according to most participants. This caused pregnant adolescents to feel apprehensive about attending antenatal clinics at the MOU.

You know what, sometimes the younger we are ne [hey]? Sometimes we scared to come to the clinic, you know, because nurses, doctors, they shout at us, [they ask us] why you having pregnancy as young you are? Why? So we are scared [to come to the MOU]. P12, 16 years old

Many participants reported that they expected to encounter rude, impatient and judgmental staff, and this made them feel fearful of coming to the MOU.

4.5.1 RUMOURS OF RUDE NURSING STAFF

Rumours of rude and judgmental nursing staff came across strongly amongst participants. As mentioned above, many participants expected to encounter staff that would chastise or judge them, and thus felt fearful of coming to the MOU. Hearing rumours about rude nurses created the expectation amongst pregnant adolescents that they would encounter rude nursing staff, and these rumours made pregnant adolescents reluctant to seek antenatal care at the MOU.

I was scared [to come to the MOU] because, um, people thought, think that the nurses are very rude, but when I came here they were nice and they assisted me. P7, 18 years old

However, once they attended services many participants found that the nursing staff were friendly, helpful and approachable.

The nurses is, the nurses is nice...and they speak nicely to you... [when the nurses are nice to me] it makes me feel more comfortable and... more relaxed, so...not so nervous anymore. P8, 19 years old

Nursing staff at the MOU also felt that many pregnant adolescents expected to find nursing staff to be rude and unfriendly, but that nursing staff at the MOU try to change these perceptions by treating all patients at the MOU with respect.

You know everybody have got that attitude that the sisters are shouting at people, so...otherwise, if you explain everything to them [pregnant adolescents], everything just goes right. P10, nurse, 57 years old

Nursing staff in this study were very aware of the negative perceptions in the community about maternal health care nurses. They recognized that this perception made pregnant adolescents reluctant to seek antenatal care and delay coming to the MOU.

Already they [pregnant adolescents] have the expectation that somebody's going to tell them what you are doing is wrong or somebody's going to point a finger at them, um, somebody's going to condemn them or judge them, you know. So they come [to the MOU] with that attitude. P11, nurse, 52 years old

4.5.2 INTERACTION WITH NURSING STAFF

Despite hearing rumours about rude nursing staff, the majority of participants were actually satisfied with their personal interaction with nursing staff.

Everyone is nice with you so...I like it. You see, they [nursing staff] handle you well, they explain to you everything so...yes, that's my experience. P6, 19 years old

Participants mostly reported that staff were friendly and understanding, and mentioned their interactions with nursing staff when asked about positive experiences at the MOU.

Actually [my experience at the MOU was] nice because they [the nursing staff], they also start, they can also start a conversation with you, they very nice, they talk to you, they ask you things that you can be comfortable to talk about. P7, 18 years old

Some participants in this study also indicated that they valued the education sessions at the MOU and provision of information by nursing staff, and that education by nursing staff was done in a manner that made pregnant adolescents feel comfortable. Education sessions took place in the waiting room in the form of talks and sometimes videos. Participants showed great interest in these talks and seemed willing to adhere to the advice given during these information sessions.

I like the explaining [education sessions], you see outside [in the waiting area] the time when you come [arrive at the MOU], I'm listening outside [to] the doctor there outside [in the waiting area], they[doctors and nurses are] talking about the baby, and the, with this baby, the first one [my first baby], maybe I wish I can take care of him [the baby] like the way the doctor he talk outside, I wish so. P12, 16 years old

Nursing staff try to remain unaffected by negative perceptions about them, and report trying to reassure pregnant adolescents that they will not be judged or mistreated by any nursing staff. Furthermore, nursing staff explained that when they speak to pregnant adolescents in a respectful manner and make them feel at ease, there is a positive relationship between nursing staff and pregnant adolescents.

That was an old perception. Because that's what I used to tell them [pregnant adolescents] if they said now 'I was scared because I know that the sisters all skel [scold]" I said no, those sisters are dead, we the ones that are here now, we don't do skel [scold] anybody, all we need is to understand each other, that's all. If I ask you something you must just tell me so that I can also understand, that's it. P10, nursing sister, 57 years old

Finding nursing staff to be approachable and friendly put the participants at ease during their visit to the MOU, and participants reported that they felt free to ask nursing staff for advice and guidance.

No, they [nursing staff] are friendly, they not rude, like the people outside [in the community] would tell me, they going to skel [reprimand me], they going to do this [reprimand] and shout at you but they not actually like that. P5, 19 years old

Nursing staff at the MOU reported that they have positive interactions with pregnant adolescents at the MOU.

[My interaction with pregnant adolescents is] very good, very good, it's very good. They ask lots of questions, and because now there is this googling stuff, you tell them this and they say Sister I google this, is this right? So when they come, some of them, they come with information, and then you just add from what they know...It's very good. You can see that they are really responsible, because you see that is responsibility to hunt for information, it's very good. P10, nursing sister, 57 years old

However, one nurse reported that she felt disconnected from many pregnant adolescents at times, and attributed this to advancing technology and numerous social stressors, such as severe poverty, exposure to abuse and child-headed homes. She also indicated that she sometimes feels a sense of hopelessness when faced with pregnant adolescents who have problems such as sexual abuse and poverty.

These kiddies [adolescents] have they've got so much happening out there [new technology and social stressors], and it's so different from our time. And fossils like myself, it's difficult to keep up with them, so when they [pregnant adolescents] come to you and they are all full of this [new technology and social stressors] it's like...the two[nurses and pregnant adolescents] doesn't meet. P11, nurse, 52 years old

Despite this, she reported that she made extra effort to treat all pregnant adolescents with dignity and compassion.

There's a lot of patience involved [in providing healthcare to pregnant adolescents] and there's a lot of teaching, and a lot of, you know, sort of, just give comfort. Unfortunately we don't have the space where we can ask mom [mothers of pregnant adolescents] to come in [to the delivery room or consultation with the pregnant adolescent] so, um, some of us [nurses] will take that role, we will slip into that role [of mother]. P11, nurse, 52 years old

Participants reported that they heard stories about rude nursing staff in the community, from friends and family members who have utilized the MOU for antenatal care in the past but actually had not experienced rude nursing staff themselves.

4.5.3 EXPECTED LONG WAITING TIMES

Participants also reported that they heard rumours about long waiting times at the MOU, which added to their reluctance to visit the MOU.

[I have heard] that... if we come here you must have patience because it's always full and you wait here the whole day and such stuff...It actually did, actually I didn't want to come here [to the MOU] when they [people in the community] said that but, when I came so it actually wasn't that bad. P2, 16 years old

Apart from expecting long waiting times, participants also expressed dissatisfaction with the actual waiting times, and reported experiencing physical discomfort due to lengthy waiting times at the MOU.

[The long waiting time] it's not nice...I mean all of us we are pregnant here outside so, we can't sit the whole time on our bums and if you stand up and then you want to sit again and it's not nice. P1, 19 years old

A common complaint about the long waiting time was experiencing back pain and hunger. Participants felt that because they are all pregnant and more prone to physical discomfort due to prolonged sitting in waiting areas, more consideration should be given to cutting waiting times.

We [patients at the MOU are] getting impatient [due to long waiting times]... moving around and...because our backs is paining, some are hungry, we came here early this morning, past six [o' clock]. P8, 19 years old

Some adolescents felt that staff at the MOU work slowly, and that staff could do more to shorten the waiting time for patients at the MOU.

I think here are a lot of people [patients] here [at the MOU], so they [staff at the MOU] can work faster...because I've been here since six o' clock this morning [the interview took place at 09h25]. P1, 19 years old

Some participants showed empathy towards staff at the MOU. Even though they reported lengthy waiting times, these participants understood the reasons for the delay in consultation and were therefore less upset by the long waiting time.

[The waiting time] was a little bit long, you know, because, but I understand because there are not lot of people [staff] here, so I understand why it was a little bit waiting and sitting and all that stuff, but it's fine. P6, 19 years old

Participants also appreciated when staff came and explained reasons for delays to them, as they then had a better understanding into why they are waiting.

4.6 POOR SELF-REPORTED KNOWLEDGE OF MATERNAL HEATLH CARE SERVICES

Many pregnant adolescents reported that they have poor knowledge of maternal health care and maternal health care service.

I knew nothing about it [antenatal clinic attendance], I didn't even know about this clinic, so it was my first time coming here. P7, 18 years old

Many pregnant adolescents also came to the MOU mainly because their mother, grandmother or friends told them to do so, and had very little knowledge about the importance of attending the MOU.

My mother told me that I should come [to the MOU] because...you need vitamins, etc. P5, 19 years old

Having poor knowledge concerning maternal health care and maternal health care services means that pregnant adolescents do not know or feel that there is a need for them to attend the MOU, resulting in poor access and utilization of maternal health care services amongst pregnant adolescents.

Some respondents reported that they had basic knowledge about maternal and reproductive health, but still viewed this knowledge as minimal.

Yes, [I] actually [know] a little bit [about maternal health]. You must eat healthy food and you must drink a lot of water, and you mustn't move...you mustn't have heavy stuff on you when you pregnant you see, so not actually much about the baby. P6, 19 years old

However, the knowledge they had about self-care during pregnancy did not include knowledge about maternal health care services and antenatal clinic attendance.

Nursing staff felt that adolescents are not utilizing reproductive health care services optimally, and that adolescents have good knowledge about reproductive health care services.

I won't say [adolescent pregnancy is due to] ignorance...I won't say because the family planning now is available to everybody, and nobody must bring the parents to come and sign for family planning...I really don't know [why the rate of adolescent pregnancy is high], because everything is available. P10, nursing sister, 57 years old.

Nursing staff thus felt that adolescents have good knowledge about reproductive and maternal health services, and that poor knowledge of reproductive and maternal health care services is not a major contributor to poor access and utilization of reproductive and maternal health care services. However, pregnant adolescents in this study reported that they had poor knowledge regarding maternal health care and SRH services.

4.7 CONCLUSION

In this study, half of pregnant adolescents had not completed their schooling. Nine out of ten pregnant adolescents reported the pregnancy as an unplanned event. Furthermore, two adolescents got married after learning that they were pregnant, and one reported getting engaged after learning that she is pregnant.

Fear of disclosing the pregnancy came across strongly as a barrier to accessing maternal health care services amongst pregnant adolescents. Many pregnant adolescents reported negative initial response to their pregnancy from parents or guardians, but this was followed by support and advice in most cases. However, in some cases, fathers tended to remain angry for longer than

mothers of pregnant adolescents did after learning about the pregnancy. Participants in this study generally had a negative perception of maternal health care services before using these services, which acted as barrier to access to maternal health care services. Many adolescents heard rumours of rude nursing staff, and feared judgment and chastisement from nursing staff. In contrast to widespread beliefs and attitudes regarding rude nurses, it appeared that the majority of participants had positive interactions with nursing staff, and reported the nursing staff to be friendly and helpful.

Many participants were worried about long waiting times, and were dissatisfied with the waiting time at the MOU, and complained of physical discomfort whilst waiting for their consultation. Many participants also reported having poor knowledge of maternal health and maternal health care services, which was a barrier to access to maternal health care services. Mothers and grandmothers were most often cited as sources of information about maternal health and

UNIVERSITY of the WESTERN CAPE

maternal health care services.

CHAPTER 5 DISCUSSION

5.1 INTRODUCTION

This chapter focuses on the findings of the study in relation to the available literature regarding adolescent pregnancy. The purpose of this study was to explore the perceptions and experiences of pregnant adolescents regarding access to maternal health services, as well as to identify barriers and facilitators to access to maternal health care services. The findings of this study showed that fear of disclosure of pregnancy is a major barrier to access to maternal health care services amongst pregnant adolescents. Pregnant adolescents feared a negative response to the pregnancy from their parents or guardians, and also experienced feelings of shock and disbelief when they first learned that they were pregnant. These feelings of shock and disbelief lead to difficulty with decision-making in terms of seeking healthcare.

Many pregnant adolescents in this study had negative perceptions about maternal health care services. Most of these perceptions were formed by hearing rumours about rude nursing staff, which made pregnant adolescents reluctant to go to the MOU. However, despite the rumours about rude nursing staff, the majority of participants reported that they experienced positive interactions with nursing staff at the MOU, contrary to their expectations. The majority of participants also reported dissatisfaction with long waiting times at the MOU. Lastly, many participants had poor knowledge with regards to maternal health care services, which contributed to poor access to maternal health care services amongst pregnant adolescents.

This chapter further unpacks the above-mentioned findings.

5.2 FEAR OF DISCLOSURE OF PREGNANCY

Many participants expressed fear and reluctance to disclose their pregnancy to their parents or guardians. This caused them to delay disclosing their pregnancy, and delay seeking health care due to fear that their pregnancy will be revealed to their parents or guardians. This is similar to findings by Chaibva et al. (2009), who found that pregnant Zimbabwean adolescents delayed seeking antenatal health care because of fear of pregnancy disclosure. Fear of pregnancy disclosure not only impedes pregnant adolescents' access to healthcare services, but also

negatively affects their access to other services, such as schooling and social support (Loxton et al., 2007; Combs-Orme, 1993).

Some participants in this study stated that they experienced feelings of guilt in response to the pregnancy, and viewed the pregnancy as a source of disappointment to their parents or guardians. This caused them to conceal the pregnancy from those around them for as long as possible. This is in-keeping with findings by Grietens et al. (2010), whose study in Burkina Faso revealed that pregnant adolescents often face feelings of shame, embarrassment and guilt when discovering that they are pregnant. Grietens et al. (2010) also found that attempts to conceal the pregnancy caused a delay in accessing maternal health care services amongst pregnant adolescents. Fear of disclosing their pregnancy is most commonly rooted in the belief that they will be judged by others and seen as immoral (Richter et al., 2006; Wiemann et al., 2005).

Ilika and Anthony (2004) found that pregnant adolescents in Nigeria who are still in school often fear the reactions of teachers and peers, and many fear punishment because of religious or cultural rules. In many communities, cultural and religious rules dictate that pregnancy should only occur after marriage. When unmarried adolescents become pregnant, they delay disclosing the pregnancy in order to conceal the fact that they have deviated from cultural or religious expectations. A culture of non-disclosure around adolescent pregnancy is thus a common occurrence in many households, with fear and shame guiding the decisions of many adolescents (Richter et al., 2006). Participants in this study at Mitchells Plain also revealed that they were concerned about the perceptions of their peers at school in response to the pregnancy, in-keeping with findings in the literature. Fear of disclosing pregnancy thus leads to delay in accessing maternal health care services amongst pregnant adolescents, as many pregnant adolescents attempt to conceal the pregnancy for as long as possible.

5.2.1 NEGATIVE INITIAL RESPONSE TO PREGNANCY FROM PARENTS

In this study, many pregnant adolescents encountered negative responses from parents and guardians when disclosing their pregnancy. These negative responses included anger and disappointment from parents. Furthermore, pregnant adolescents in this study often anticipated this negative response, which also contributed to fear of disclosing the pregnancy. These findings are supported by a study done by Van Zyl et al. (2015), who found that pregnant adolescents in

the Western Cape are often scolded and expelled from home when the pregnancy is revealed to their parents or guardians.

However, participants in this current study reported that the initial negative response was often followed by support and advice from parents and guardians. Mturi (2015) had similar findings in a study in the North West Province of South Africa. The initial anger that parents felt was short-lived and in keeping with findings in this current study, pregnant adolescents cited their mothers as pivotal in providing support and guidance through their pregnancy experience (Mturi, 2015). The manner in which parents respond to the pregnancy greatly influences the manner in which pregnant adolescents manages their pregnancy (Mumah et al., 2014). This includes the decision on when to seek health care for pregnancy, and self-care practices during pregnancy.

A possible reason for the initial negative response to the pregnancy by parents and guardians in this current study in the Mitchells Plain sub-district is that parents fear the negative impact that early pregnancy may have on future schooling and career prospects for the adolescent (Singh and Hamid, 2015). Early adolescent pregnancy in South Africa has been found to have negative implications on fulfilment of scholastic and career goals for young mothers, as well as placing financial strain on households. Many households in Mitchells Plain face financial hardship, and an unplanned pregnancy puts further financial strain on families, which contributes to parents and guardians responding negatively to initial news of an unplanned adolescent pregnancy in the household. However, Singh and Hamid (2015) suggest that becoming a mother is held in high regard in many cultures in South Africa, and families therefore tend to become supportive after the initial negative response.

Most of the participants in this study at Mitchells Plain MOU indicated that their parents or guardians offered support and guidance after the initial negative response. Some participants also reported that their parents strongly encouraged them to continue with schooling. Singh and Hamid (2015) as well as Van Zyl et al. (2015) found that adolescents who receive support from parents during their pregnancy and with child-rearing tend to have a more positive outlook on their future, and use their pregnancy experience as motivation to improve their lives for themselves and their child. The support and guidance of parents and guardians is thus important for positive outcomes for the adolescent mother and her baby.

5.3 FEELINGS OF SHOCK AND DISBELIEF IN RESPONSE TO PREGNANCY

Many adolescents in this study at Mitchells Plain MOU expressed feeling shocked when they discovered that they were pregnant, which contributed to a delay in seeking antenatal care. This is in keeping with findings by Atekyereza and Mubiru (2014) as well as Mturi (2015), who found that when pregnancy is unplanned, women often feel overwhelmed and shocked when pregnancy is confirmed. When pregnancy is deemed as unacceptable by society, such as adolescent pregnancy, the feelings of shock and disbelief are often amplified. The feelings of shock and disbelief means that pregnant adolescents delay seeking early antenatal care, as they first try to process the array of emotions they feel whilst simultaneously trying to avoid judgment from family and society (Singh & Hamid, 2015).

The shock and fear that pregnant adolescents experience not only prevents them from accessing health care, but often also causes them to isolate themselves from family and friends (De Villiers & Kekesi, 2004). The difficulty pregnant adolescents' face in coming to terms with an unplanned pregnancy may thus greatly hinder the amount of support they receive from the health care workers, family and friends. In many cases, the pregnant adolescent receives assistance and advice to attend antenatal clinic as soon as the family is made aware of the pregnancy (Pell et al., 2013). Thus, when feelings of shock and disbelief cause pregnant adolescents to isolate themselves from family, it causes a delay in access to antenatal care.

5.4 PERCEPTIONS AND EXPERIENCES OF MATERNAL HEALTH CARE SERVICES

For many women in low and middle-income countries, a negative experience of maternal health care services is a barrier to access to maternal health care (Finlayson & Downe, 2013). Pregnant adolescents in particular have the expectation that they will chastised by healthcare workers when seeking health care for sexual and reproductive health (SRH) care needs (Kennedy et al., 2013). Negative perception of maternal health care services is thus a significant deterrent to early access to maternal health care services. These findings are in keeping with this study at Mitchells Plain MOU, where numerous participants expressed that they expected to be scolded by nursing staff for being pregnant at a young age, and therefore feared seeking antenatal care. This caused participants to delay seeking antenatal care.

Some participants in this study at Mitchells Plain MOU expressed that they felt uncomfortable attending the MOU with older pregnant women. Similarly, adolescent participants in a study by Ehlers et al. (2000) revealed that they felt uncomfortable at the prospect of attending clinics with older pregnant women, and believed that older women attending the clinic may judge them.

One participant in this study at Mitchells Plain MOU reported that she had not yet disclosed the pregnancy to her parents, and that she was reluctant to come to the MOU out of fear that someone at the MOU may reveal the pregnancy to her parents. The literature shows that pregnant adolescents frequently report fears regarding confidentiality when accessing maternal health care services (Agampodi et al., 2008). They hold the perception that accessing maternal health care services will lead to their pregnancy status being revealed to their parents either by health care workers (HCW's), or through being recognized by community members. Concerns regarding confidentiality of maternal health care services thus act as a barrier to access to maternal health care services amongst pregnant adolescents.

When adolescents hold negative perceptions of maternal health care services, they are more likely to have poor attendance at antenatal clinics (Gross et al, 2012). As seen in this study at Mitchells Plain MOU, an expected negative experience of maternal health care services causes many pregnant adolescents to delay accessing maternal health care services.

5.4.1 RUMOURS OF RUDE NURSING STAFF

Many participants in this study at reported that they heard rumours about rude nursing staff at Mitchells Plain MOU. These rumours were heard in the community, or from friends and family. Silal et al. (2012) also found that in South Africa, word of mouth regarding judgment and harsh treatment by nurses towards pregnant adolescents served as a barrier to access to maternal health care services amongst adolescents.

Similarly, in a study by Bohren et al. (2017), it was found that many Nigerian women experienced mistreatment by nursing staff at maternal health care facilities, and pregnant adolescents were perceived to be particularly susceptible to mistreatment from nursing staff. In some cases, nursing staff treat adolescents harshly as means of discouraging sexual activity amongst adolescents (Wood & Jewkes, 2006). Nursing staff that treat adolescents who seek SRH

care with disrespect are often influenced by their personal belief that adolescents should not engage in sexual activity (Jonas et al., 2017). Adolescents have become aware of this treatment at health care facilities, and thus expect the same treatment when accessing maternal health care services.

Many pregnant adolescents who participated in this study at Mitchells Plain MOU have had very little experience with maternal health care services. This lack of experience of maternal health care services means that pregnant adolescents easily believe negative information they receive about the treatment they might encounter maternal health care facilities (Loxton et al., 2007). Attitudes of HCW's are thus an important influence on how easily pregnant adolescents feel they can access maternal health care services (Phafoli et al., 2007).

5.4.2 INTERACTION WITH NURSING STAFF

Despite rumours of rude nursing staff and expecting to be treated poorly at Mitchells Plain MOU, many pregnant adolescents in this study reported positive interaction with nursing staff. Participants described nursing staff as friendly and approachable, and reported that the friendly attitude of nurses made them feel at ease. Phafoli et al. (2007) had similar findings, with numerous pregnant adolescents in their study reporting that were happy with the manner in which nursing staff treated them. Positive attitudes of nursing staff towards adolescents are thought to encourage access to SRH services amongst adolescents (Jonas et al., 2017). However, in a study in South Africa, Alli et al. (2012) found that many adolescents still reported negative interactions with HCW's when seeking reproductive health care, though some adolescents were happy with the services they received.

Nursing staff at Mitchells Plain MOU reported that they had mostly positive interactions with pregnant adolescents at the MOU. An Ethiopian study by Tilahun et al. (2012) supports these findings. They found that many health workers had positive interactions with adolescents seeking SRH care services, and a smaller number of health care workers had a negative outlook on the provision of SRH care services to adolescents. Even though most of the HCW's had good interactions with adolescents, the negative actions of the smaller group of HCW's in their study acted as a barrier to access to SRH care amongst adolescents. Thus, even though many adolescents as well as HCW's report positive interactions, the negative experiences that a smaller

number of adolescents and HCW's have in the provision of SRH care services seem to filter into communities through word of mouth, creating the expectation amongst adolescents that they will be mistreated at health care facilities.

Bowser and Hill (2010) conducted study to investigate the occurrence of maltreatment of women at maternal health care centres in various countries. Contrary to findings in this study at Mitchells Plain MOU, Bowser and Hill (2010) found that HCW's frequently mistreat pregnant women and teenage mothers in particular. The authors suggest that mistreatment of vulnerable groups of pregnant women, such as pregnant adolescents, has become accepted practice in many institutions. Many pregnant adolescents thus do not report negative interactions or experiences at maternal health care facilities.

5.4.3 EXPECTED LONG WAITING TIMES

Many participants in this study expressed dissatisfaction with waiting times at Mitchells Plain MOU. Furthermore, participants had heard about extremely long waiting times before coming to the MOU, and this caused many participants to be reluctant to go to the MOU. This is in keeping with findings by Resty et al. (2011), who conducted a mixed-methods study in Uganda to determine barriers and facilitators to utilization of maternal health care services amongst pregnant adolescents. They found that long waiting times was a barrier to access to maternal health care amongst pregnant adolescents.

Adolescents in the Eastern Cape, South Africa, also mentioned long waiting as a barrier to access to SRH services (Mayeye et al., 2010). When faced with the prospect of long waiting times, adolescents often delay accessing reproductive health care services, and will access health care services according whether or not they consider their condition to be serious enough to warrant attention from a health care professional (Onokerhoraye & Dudu, 2017). This means that pregnant adolescents may delay coming to the MOU due to expected long waiting times, and may feel that the delay in access to the MOU is acceptable because they do not feel physically ill. The prospect of long waiting times means that pregnant adolescents are not able to continue with other responsibilities they made have for the day, such as school, housework or other work responsibilities (Hokororo et al., 2015). This then further motivates pregnant adolescents to

either delay the first visit to the MOU, or default future appointments, particularly if they feel physically healthy and do not feel the need for consultation with a health care professional.

5.5 POOR SELF REPORTED KNOWLEDGE OF MATERNAL HEALTH CARE SERVICES

Participants in this study at Mitchells Plain MOU reported that they have very little knowledge about maternal healthcare services. Many reported that they received advice from their mother, grandmother or partner to attend the MOU and did not realize that early attendance at the MOU was important until they received advice to attend. Chaibva et al. (2009) also found that lack of knowledge about maternal health care services was a barrier to access to antenatal care for pregnant adolescents. Lack of knowledge concerning maternal health services is common amongst young pregnant women (Loxton et al., 2009). According to Loxton et al. (2009), the gap in knowledge includes knowing when to consult maternal health care centres, as well as knowing what services are available to be accessed.

Lack of knowledge about maternal health care services has an impact on the adolescent mother as well as her child (Combs-Orme, 1993). Often, adolescent mothers also show poor utilization of services after childbirth due to lack of knowledge of which healthcare services her baby now needs. Women who have good knowledge about maternal health care services tend to utilize maternal health care services earlier and as recommended, compared to those with poorer knowledge regarding maternal health care services (Nisar & White, 2003). Knowledge regarding maternal health care services is thus an important determining factor for access and utilization of maternal health care services, with poor knowledge acting as a barrier to access to maternal health care services.

Many participants in this study at Mitchells Plain MOU cited their mother or grandmother as sources of information concerning antenatal care and utilization of maternal health care services. Very often, mothers, grandmothers or aunts advise young girls about SRH matters such as menstruation and pregnancy prevention (Henry & Fayorsey, 2002). However, in most cases, it appears that education regarding antenatal care is only done after pregnancy disclosure. This could be because adolescent pregnancy is deemed unacceptable in many cultures and settings, or

that parents and guardians do not feel comfortable discussing pregnancy and antenatal care with adolescent girls.

5.6 LIMITATIONS

A limitation of this study is that adolescents who have not accessed the MOU were excluded. Finding those who are not accessing the MOU is extremely difficult as there was no feasible manner within the period and scope of a mini-thesis to find those who should be accessing the MOU but are not, making exclusion necessary. This means that the perceptions of pregnant adolescents who have not accessed the MOU are not known in this study. Sampling only those that have accessed the MOU means that they may be more likely to give positive feedback with regards to family support, access to services and healthcare workers.

Of the pregnant adolescents below the age of 18, only those with parental permission were included in this study, so the perspectives of pregnant adolescents below the age of 18 who did not have parental permission are unknown. Pregnant adolescents under the age of 16 years were excluded from this study due to the fact that sexual intercourse is illegal for anyone under the age of 16 years of age. The views of those below the age of 16 are thus unknown in this study, as participation will mean exploring legal issues regarding sexual activity in this group. Discussing the legal issues about their sexual history may cause undue emotional or psychological stress for participants under the age of 16. It is therefore in the best interest of those under 16 years old to be excluded.

Another limitation of this study is that I have only included participants who speak English and Afrikaans. The researcher herself funded this research, therefore budgetary constraints did not allow for the use of an interpreter or translator. As the researcher speaks only English and Afrikaans, it was not feasible to conduct the interviews in any other languages. This means that the experiences of pregnant adolescents utilizing the MOU but do not speak English or Afrikaans is not accounted for in this research.

Conducting the research at the MOU may also mean that participants could be reluctant to give a negative account of their experience at the MOU. Participants are still in the same vicinity as the nursing staff at the MOU during the interviews, albeit in a separate cubicle and out of immediate

sight. This may make it difficult for participants to give negative feedback about the MOU or staff at the MOU.

Nursing staff at the MOU may also be unlikely to report their own negative behaviour, which means that they could give a bias account of their interaction with pregnant adolescents.

The role of family planning and contraception, as well as the role of fathers in both planned and unplanned pregnancies amongst adolescents are relevant areas to explore in pregnancies amongst adolescents. However, the limited time and scope for this study did not allow for exploration of these topics.

5.7 SUMMARY

The findings of this research correlate with findings from the literature. Barriers to access to maternal health care services amongst pregnant adolescents include fear of disclosing their pregnancy, negative reactions to the pregnancy, as well as feelings of shock and disbelief. Many pregnant adolescents have negative perceptions about maternal health care services, and expect to encounter harsh treatment and judgement from nursing staff. These negative perceptions cause many pregnant adolescents to delay accessing the MOU. However, many pregnant adolescents report positive interaction with nursing staff, in spite of the negative perceptions they had about the MOU. Long waiting times served as another barrier to access to maternal health care services amongst pregnant adolescents, and poor knowledge about maternal health care services also inhibits early access to maternal health care services amongst pregnant adolescents.

6.1 CONCLUSION

The aim of this study was to explore the barriers to access for maternal health care amongst pregnant adolescents in the Mitchells Plain sub-district. This study has shown that pregnant adolescents in the Mitchells Plain sub-district often fear disclosing the pregnancy to their parents or caregivers, and this has been identified as a major barrier to access to maternal health care services. Furthermore, adolescents often anticipate a negative response to the pregnancy from their parents or caregivers, and therefore try to hide the pregnancy for as long as possible. While trying to hide the pregnancy, pregnant adolescents avoid seeking care at maternal health care centres, because they then run the risk of their parents or guardians learning about the pregnancy if they attend antenatal clinics. Results of this study showed that most adolescents only started visiting the MOU after disclosing the pregnancy to parents or caregivers. Many pregnant adolescents at Mitchells Plain MOU reported that a negative response to the pregnancy from parents and guardians was followed by support and guidance from parents, which facilitated access to maternal health care services amongst pregnant adolescents. Advice from pregnant adolescents' mothers or grandmothers to attend the MOU was also a facilitator to access to maternal health care services amongst pregnant adolescents in the Mitchells Plain sub-district.

Many pregnant adolescents in this study had a negative perception of maternal health care services at Mitchells Plain MOU, which made them reluctant to attend the MOU. These perceptions were often created due to rumours that pregnant adolescents heard in the Mitchells Plain community about rude nursing staff prior to going to the MOU. However, the majority of adolescents in this study had positive experiences in terms of interaction with nursing staff at the MOU despite the negative rumours they heard. Nursing staff in this study were aware of the negative perceptions of nursing staff and services at the MOU, and had a desire to change that perception amongst pregnant adolescents. Nursing staff also reported that they had positive interactions with pregnant adolescents, but also felt helpless at times due to the difficult situations endured by pregnant adolescents in the Mitchells Plain community, such as poverty.

Most participants in this study were unhappy with the waiting time at Mitchells Plain MOU, which acts as a barrier to access to maternal health care services, and may contribute to pregnant adolescents defaulting future appointments at the MOU. Lack of knowledge about maternal health care services is another barrier to maternal health care services amongst pregnant adolescents in the Mitchells Plain sub-district. Very often, adolescents in this study received advice from their mother or grandmother to attend the MOU after disclosing the pregnancy, but had little or no knowledge of maternal health care services before receiving advice from their mother or grandmother. This gap in knowledge contributed to poor antenatal attendance at Mitchells Plain MOU prior to disclosure of the pregnancy.

Adolescent access to maternal health care services the Mitchells Plain sub-district is thus a complex issue. The perceptions, beliefs and attitudes that adolescents have towards maternal health care services are shaped by societal norms and expectations about adolescent childbearing, as well as rumours they have heard in the community concerning how pregnant adolescents are treated by healthcare workers. Negative perceptions, beliefs and attitudes as well as lack of knowledge about maternal health care thus hinder access to maternal health care services amongst pregnant adolescents.

UNIVERSITY of the

6.2 RECOMMENDATIONS

Based on the findings of this research, it is clear that early pregnancy disclosure amongst pregnant adolescents should be encouraged. This will require safe spaces to be created in which adolescents can disclose their pregnancy and receive assistance to access maternal health care services. This should include inter-sectoral collaboration between the Department of Health, Department of Education as well as social welfare to ensure that pregnant adolescents receive the assistance they need in all spheres, and receive assistance to disclose the pregnancy to their parents or guardians if required. There should be a clear and well-functioning referral pathway between schools and local clinics, so that learners who suspect that they are pregnant can receive assistance from schools to get tested at local clinics, and in this way pregnant adolescents can access maternal health care services early in pregnancy. Schools should also have a clear referral pathway to social welfare departments to mediate in cases where learners are afraid to disclose the pregnancy to their parents or guardians. Parents and guardians should also be given tools on

how to cope with an unplanned adolescent pregnancy in the family, in order to better assist pregnant adolescents. This could be provided by the Department of Social Development. This interrelated pathway will require collaboration between the Department of Health, Department of Social Development as well as the Department of Education.

It may be beneficial to include education regarding antenatal care and maternal health care services in high schools' life orientation curriculum, so that all adolescents understand the importance of antenatal care, and know where and how to access maternal health care services. As part of inter-sectoral collaboration, representatives from the Department of Health may provide some of these education sessions to learners, so that the perception that adolescents are mistreated when seeking reproductive health care is addressed.

The South African Department of Health started the Integrated School Health Policy (ISHP) in 2012, which includes collaboration between the education, health and social development sectors to improve the health of school-going learners through all phases of primary and secondary education (Department of Health & Department of Basic Education, 2012). This includes SRH services, referral systems and collaboration between the relevant departments in formulating schools' life orientation curriculum. However, the ISHP has been gradually phased in since 2012, and thus is not uniformly and consistently implemented. The implementation of the ISHP thus needs to be leveraged, particularly in areas with high rates of adolescent pregnancy to ensure that inter-sectoral collaboration in the management if adolescent pregnancy.

To further address the perceptions of maternal health care services amongst adolescents, it is important that maternal health nurses receive training and education regarding the health care needs of pregnant adolescents. This training should include sensitization to the difficulties faced by pregnant adolescents when accessing maternal health care services, perceptions of maternal health care services amongst adolescents, as well as improving communication skills and professionalism towards pregnant adolescents. This may assist in bridging the communication gap that some nurses may feel when providing services to pregnant adolescents. Further research into adolescent access to maternal health care services is recommended. Further research into the role of family planning and contraception, as well as the role of fathers in adolescent pregnancy is also recommended.

REFERENCES

AbouZahr, C. (2013). *Trends and projections in mortality and morbidity*. WHO International Conference on Population Development, Women's Meeting. Available at https://www.unfpa.org/sites/default/files/resource-pdf/Overview.pdf (accessed 11/7/2015).

Ahorlu, C.K., Pfeiffer, C., Obrist, B. (2015). *Socio-cultural and economic factors influencing adolescents' resilience against the threat of teenage pregnancy: A cross-sectional survey in Accra, Ghana*. Reproductive Health, 12 (117). Available at https://reproductive-health-journal.biomedcentral.com/articles/10.1186/s12978-015-0113-9 (accessed 8/2/2017).

Agampodi, S.B., Agampodi, T.C., Piyaseeli, U.K.D. (2008). *Adolescents perception of reproductive health care services in Sri Lanka*. BMC Health Services Research, 8 (98). Available at http://www.biomedcentral.com/1472-6963/8/98 (accessed 7/5/2017).

Alli, F., Maharaj, P., Vawda, M.Y. (2012). *Interpersonal relations between health care workers and young clients: barriers to accessing sexual and reproductive health care.*Journal of Community Health. Available at http://www.hsrc.ac.za/en/research-outputs/ktree-doc/12020 (accessed 6/5/2017).

Althabe, F., Moore, J., Gibbons, L., Berrueta, M., Goudar, S., Chomba, E., Derman, R.J., Patel, A., Saleem, S., Pasha, O., Esamai, F., Garces, A., Liechty, E.A., Hambidge, K.M., Krebs, N.F., Hibberd P.L., Goldenberg, R.L., Koso-Thomas, M., Carlo, W.A., Cafferata, M.L., Buekens, P., McClure, E.M. (2015). *Adverse maternal and perinatal outcomes in adolescent pregnancies: The Global Network's Maternal Newborn Health Registry study*. Reproductive Health, 12(Suppl 2) S8. Available at http://www.reproductive-health-journal.com/content/12/S2/S8 (accessed 16/7/2015)

Atekyereza, P.R., Mubiru, K. (2014). *Influence of pregnancy perceptions on patterns of seeking antenatal care among young women in reproductive age of Masaka District,*

Uganda. Tanzania Journal of Health Research, 16 (4), 312-321. Available at https://www.ncbi.nlm.nih.gov/pubmed/26891521 (accessed 6/5/2017).

Atuyambe, L., Mirembe, F., Tumwesigye, N.M., Annika, J., Kirumira, E.K., Faxelid, E. (2008). *Adolescent and adult first time mothers' health seeking practices during pregnancy and early motherhood in Wakiso district, central Uganda*. Reproductive Health, 5 (13). Available at http://www.reproductive-health-journal.com/content/5/1/13 (accessed 10/7/2015).

Bohren, M.A., Vogel, J.P., Tunçalp, Ö., Fawole, B., Titiloye, M.A., Olutayo, A.O., Ogunlade, M., Oyeniran, A.A., Osunsan, O.R., Metiboba, L., Idris, H.A., Alu, F.E., Oladapo, O.T., Gülmezoglu, A.M., Hindin, M.J. (2017). *Mistreatment of women during childbirth in Abuja, Nigeria: a qualitative study on perceptions and experiences of women and healthcare providers*. Reproductive Health, 14 (9). Available at https://reproductive-health-journal.biomedcentral.com/articles/10.1186/s12978-016-0265-2 (accessed 7/5/2017).

Bowser, D., Hill, K. (2010). *Exploring evidence for disrespect and abuse in facility-based childbirth. Report of a landscape analysis*. USAID TRAction Project. Available at http://www.tractionproject.org/resources/access-skilled-care-respectful-maternal-care/exploring-evidence-disrespect-and-abuse (accessed 6/5/2017).

Brahmbhatt, H., Kågesten, A., Emerson, M., Decker, M.R., Olumide, A.O., Ojengbede, O., Lou, C., Sonenstein, F.L., Blum, R.W., Delany-Moretlwe, S. (2014). *Prevalence and determinants of adolescent pregnancy in urban disadvantaged settings across five cities*. Journal of Adolescent Health, 55, S48-S57. Available at www.jahonline.org (accessed 27/10/2016).

Burack, R. (2000). Young adolescents 'attitudes towards general practitioners and their provision of sexual health care. British Journal of General Practice, 50, 550-554.

Chaibva, C.N., Roos, J.H., Ehlers, V.J. (2009). *Adolescents mothers' non-utilisation of antenatal care services in Bulawayo*, *Zimbabwe*. Curationis, 32 (3), 14-21.

Chandra-Mouli, V., Camacho, A.V., Michaud, P.A. (2013). WHO guidelines on preventing early pregnancy and poor reproductive outcomes among adolescents in developing countries. Journal of Adolescent Health, 52, 517-522.

Chandra-Mouli, V., McCarraher, D.R., Phillips, S.J., Williamson, N.E., Hainsworth, G. (2014). *Contraception for adolescents in low and middle income countries: needs, barriers and access*. Reproductive Health, 11 (1). Available at http://www.reproductive-health-journal.com/content/11/1/1 (accessed 8/8/2015).

Chigona, A., Chetty, R. (2007). *Girls' education in South Africa: Special consideration to teen mothers as learners*. Journal of Education for International Development, 3 (1). Available at

https://pdfs.semanticscholar.org/31af/180887c35164bbada29d6e1e56d8a81e47db.pdf (accessed 17/1/2017).

Choy, L. (2014). The Strengths and Weaknesses of Research Methodology: Comparison and Complimentary between Qualitative and Quantitative Approaches. IOSR Journal of Humanities and Social Science, 19 (4) Ver. III, 9-104. Available online: http://iosrjournals.org/iosr-jhss/papers/Vol19-issue4/Version-3/N0194399104.pdf (Accessed 26/9/2015).

Christofides, N., Jewkes, R., Dunkle, K., Nduna, M., Shai, N., Sterk, C. (2014). *Early adolescent pregnancy increases risk of incident HIV infection in the Eastern Cape, South Africa: a longitudinal study*. Journal of the International AIDS Society, 17(18585). Available at http://www.jiasociety.org/index.php/jias/article/view/18585 (accessed 12/7/2017).

City of Cape Town. (2013). City of Cape Town-2011 Census Suburb Mitchells Plain. Available at

https://www.capetown.gov.za/en/stats/2011CensusSuburbs/2011_Census_CT_Suburb_Mitchells_Plain_Profile.pdf. (Accessed 15/8/2015)

Coinco, E. (2010). *A glimpse into the world of teenage pregnancy in Sierra Leone*. UNICEF Final Report. Available at

https://www.humanitarianresponse.info/sites/www.humanitarianresponse.info/files/assessments/teenage_pregnancy_report_unicef_final_aug_2010_2.pdf (accessed 7/2/2017).

Combs-Orme, T. (1993). *Health effects of adolescent pregnancy: implications for social workers*. Social Work Publications and Other Works, 344-354. Available at http://trace.tennessee.edu/utk_socipubs/22 (accessed 7/5/2017).

Copland, R.J., Denny, S.J., Robinson, E.M., Crengle, S., Ameratunga, S., Dixon, R. (2011). Self-reported pregnancy and access to primary health care among sexually experienced New Zealand high school students. Journal of Adolescent Health, 49, 518-524.

Creswell, J.W., Miller, D.L. (2000). *Determining Validity in Qualitative Inquiry*. Theory into Practice, 39(3), 122 – 130.

Dellar, R., Waxman, A., Abdool Karim, Q. (2015). *Understanding and responding to HIV risk in young South African women: Clinical perspectives*. South African Medical Journal, 105 (11). Available at http://www.samj.org.za/index.php/samj/article/view/10099 (accessed 8/1/2017).

UNIVERSITY of the

Department of Health South Africa. (2017). *Demographic and Health Survey 2016. Key Indicators Report*. Statistics South Africa. Available online: http://www.statssa.gov.za/publications/Report%2003-00-09/Report%2003-00-092016.pdf (accessed 28/5/2017).

Department of Health., Department of Basic Education. (2012). *Integrated School Health Policy*. Available at https://www.health-e.org.za/wp-content/uploads/2013/10/Integrated_School_Health_Policy.pdf (accessed 8/6/2017).

Department of Provincial and Local Government South Africa. (2011). Mitchell's Plain Nodal Economic Development Profile. Available online: https://mitchellsplain.files.wordpress.com/2011/07/mitchells_20plain_20narrative.pdf (accessed 15/7/2016).

De Villiers, F.P.R., Kekesi, J. (2004). *Social interaction of teenage mothers during and after their pregnancy*. South African Family Practice, 46 (2), 21-24.

Dookie, S., Singh, S. (2012). *Primary health services at district level in South Africa: a critique of the primary health care approach*. BMC Family Practice 13 (67). Available at http://www.biomedcentral.com/1471-2296/13/67 (accessed 5/10/2015).

Draper, A.K. (2004). Workshop on 'developing qualitative research method skills: analyzing and applying your results'. The principles and application of qualitative research. Proceedings of the Nutrition Society, 63, 641-646

Durojaye, E. (2009). Realizing access to sexual health information and services for adolescents through the protocol to the African Charter on the rights of women. Washington and Lee Journal of Civil Rights and Social Justice, 16 (1), 135-172.

Ehlers, V.J., Maja, T., Sellers, E., Gololo, M. (2000). *Adolescent Mothers' utilisation of Reproductive Health Services in the Gauteng Province of The Republic Of South Africa*. Curationis, 43-53. Available at http://www.curationis.org.za/index.php/curationis/article/viewFile/695/633 (accessed 15/7/2016).

Ekong, I. (2016). Adequacy of adolescent healthcare services available for adolescent girls in a Southern Nigerian environment. International Research Journal of Medicine and Biomedical Sciences, 1(2), 23-28. Available at http://www.journalissues.org/IRJMBS/ (accessed 8/2/2017).

Erulkar, A.S., Onoka, C.J., Phiri, A. (2005). What is youth-friendly? Adolescents' preferences for reproductive health services in Kenya and Zimbabwe. African Journal of Reproductive Health, 9 (3), 51-58.

Fatti, G., Shaikh, N., Eley, B., Jackson, D., Grimwood, A. (2014). Adolescent and young pregnant women at increased risk of mother-to-child transmission of HIV and poorer maternal and infant health outcomes: A cohort study at public facilities in the Nelson Mandela Bay Metropolitan district, Eastern Cape, South Africa. South African Medical Journal, 104 (12), 874-880.

Finlayson, K., Downe, S. (2013). Why do women not use antenatal services in low-and-middle-income countries? A meta-synthesis of qualitative studies. PLoS Medicine 10 (1). Available at

http://journals.plos.org/plosmedicine/article?id=10.1371/journal.pmed.1001373 (accessed 6/5/2017).

Flanagan, A., Lince, N., Durao de Menezes, I., Mdlopane, L. (2013). *Teen pregnancy in South Africa: A literature review examining contributing factors and unique interventions*. Ibis Reproductive Health. Available at http://www.sexrightsafrica.net/wp-content/uploads/2016/11/2013.02.25-Teenage-Pregnancy-South-Africa-FINAL2-copy.pdf (accessed 8/1/2017).

Fraser, A., Brockert, J., Ward, R. (1995). *Association of young maternal age with adverse reproductive outcomes*. The New England Journal of Medicine, 32 (17), 1113-1117. Available online: www.nejm.org (Accessed 10/7/2015).

Gevers, A., Mathews, C., Cupp, P., Russell, M., Jewkes, R. (2013). *Illegal yet developmentally normative: a descriptive analysis of young, urban adolescents' dating and sexual behaviour in Cape Town, South Africa*. BMC International Health and Human Rights, 13 (31). Available at http://www.biomedcentral.com/1472-698X/13/31 (accessed 16/7/2015)

Godia, P.M., Olenja, J.M., Hofman, J.J., van den Broek, N. (2014). *Young people's perception of sexual and reproductive health services in Kenya*. BMC Health Services Research, 14 (172). Available at http://www.biomedcentral.com/1472-6963/14/172 (accessed 8/2/2017).

Grietens, K.P., Gies, S., Coulibaly, S.O., Ky, C., Somda, J., Toomer, E., Ribera, J.M., D'Allesandro, U. (2010). *Bottlenecks for high coverage of intermittent preventive treatment in pregnancy: The case of adolescent pregnancies in rural Burkino Faso*. PLoS ONE 5(8). Available at www.plosone.org (accessed 6/5/2017).

Groenewald, P., Bradshaw, D., Daniels, J., Matzopoulos, R., Bourne, D., Blease, D.,

Zinyaktira, N., Naledi, N.T. (2008). *Cause of death and premature mortality in Cape Town,* 2001-2006. Cape Town: South African Medical Research Council, 2008.

Gross, K., Alba, S., Glass, T.R., Armstrong Schellenberg, J., Obrist, B. (2012). *Timing of antenatal care for adolescent and adult women in south-eastern Tanzania*. BMC Pregnancy and Childbirth, 12 (16). Available at http://www.biomedcentral.com/1471-2393/12/16 (accessed 17/1/2017).

Haider, A. (2008). *Adolescents under international law: Autonomy as the key to reproductive health*. William and Mary Journal of Women and the Law, 14 (3), 605-633. Available at http://scholarship.law.wm.edu/wmjowl/vol14/iss3/5 (26/10/2016)

Haque, M. (2002). Public sector district health system: Is seven to ten minutes enough for patient management? South African Family Practice, 25 (3), 13-17.

Henry, R., Fayorsey C. (2002). *Coping with pregnancy: Experiences of adolescents in Ga Mashi, Accra*. Calverton, Maryland, USA: ORC Macro. Available at http://pdf.usaid.gov/pdf_docs/PNACU401.pdf (accessed 12/1/2017).

Hill, L., Maman, S., Groves, A., Moodley, D. (2015). Social support among HIV-positive and HIV-negative adolescents in Umlazi, South Africa: changes in family and partner relationship during pregnancy and the postpartum period. BMC Pregnancy and childbirth, 15 (117). Available at https://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/s12884-015-0542-z (accessed 12/1/2017).

Hoga, L.A.K., Borges, A.L.V., Alvares, R.E.C. (2009). *Teen pregnancy: values and reactions of family members*. Acta Paul Enferm, 22(6), 779-785.

Hokororo, A., Kihunrwa, A.F., Kalluvya, S., Changalucha, J., Fitzgerald, D.W., Downs, J.A. (2015). *Barriers to access reproductive healthcare for pregnant adolescent girls: a qualitative study in Tanzania*. Acta Paediatr, 104 (12). Available at https://www.ncbi.nlm.nih.gov/pubmed/25473729 (accessed 6/5/2017).

Hoopes, A.J., Chandra-Mouli, V., Steyn, P., Shilubane, T., Pleaner, M. (2015). *An analysis of adolescent content in South Africa's contraception policy using a human rights framework*. Journal of Adolescent Health, 57, 617-623.

Horwood, C., Butler, L.M., Haskins, L., Phakathi, S., Rollins, N. (2013). *HIV-infected mothers and their infants: Low coverage of HIV services and high risk of HIV transmission in Kwazulu-Natal, South Africa. PLoS ONE 8(9)*. Available at http://journals.plos.org/plosone/article?id=10.1371/journal.pone.0074568 (accessed 20/5/2017).

Ilika, A., Anthony, I. (2004). *Unintended pregnancy among unmarried adolescents and young women in Anambra State, South East Nigeria*. African Journal of Reproductive Health, 8(3), 92-102.

Jack, S.M. (2006). *Utility of qualitative research findings in evidence-based public health practice*. Public Health Nursing, 23 (3), 277-283.

James, S., Rall, N., Strümpher, J. (2012). *Perceptions of pregnant adolescents with regard to the antenatal clinic environment*. Curationis 35 (1). Available at http://dx.doi.org/10.4102/ curationis.v35i1.43 (accessed 6/2/2017).

Jonas, K., Crutzen, R., van den Borne, B., Reddy, P. (2017). *Healthcare workers'* behaviors and personal determinants associated with providing adequate sexual and reproductive healthcare services in sub-Saharan Africa: a systematic review. BMC Pregnancy and Childbirth, 17 (86). Available at https://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/s12884-017-1268-x (accessed 6/5/2017).

Kanku, T., Mash, R. (2010). Attitudes, perceptions and understanding amongst adolescents regarding teenage pregnancy, sexuality and contraception in Taung. South African Family Practice 2010, 52 (6), 563-572.

Karra, M., Lee, M. (2012). *Human capital consequences of teenage childbearing in South Africa*. Population Reference Bureau. Available at www.prb.org (accessed 6/5/2017).

Kennedy, E.C., Bulu, S., Harris, J., Humphreys, D., Malverus, J., Gray, N.J. (2013). "Be kind to young people so they feel at home": a qualitative study of adolescents' and service providers' perceptions of youth-friendly sexual and reproductive health services in Vanuatu. BMC Health Services Research 13 (455). Available at http://www.biomedcentral.com/1472-6963/13/455 (accessed on 7/5/2017).

Kibombo, R., Neema, S., Moore, A.M., Ahmed, F.H. (2008). *Adults' perceptions of adolescents' sexual and reproductive health: qualitative evidence from Uganda*. Guttmacher Institute, 35. Available at www.guttmacher.org. (accessed 7/2/2017)

Klein, J.D. (2005). *Adolescent Pregnancy: Current Trends and Issues*. American Academy of Paediatrics, 116 (1), 281-286. Available at: www.pediatrics.aappublications.org (accessed 19/9/2015).

Lesedi, C., Hoque, M.E., Ntuli-Ngcobo, B. (2011). Youth's perception towards sexual and reproductive health services at Family Welfare Association Centers in Botswana. Journal of Social Sciences, 28(2), 137-143.

Lenters, L., Hackett, K., Barwick, M., Zlotkin, S. (2015). *Perceptions and experiences of adolescent mothers accessing antenatal care services in Volta and Eastern Regions, Ghana, and Singida and Iramba Regions, Tanzania*. Muskoka Initiative Consortium. Available at http://www.canwach.ca/wp-content/uploads/2015/05/Adolescent-Mothers-and-ANC-Final-Report.pdf (accessed 7/2/2017).

Levesque, J., Harris, M., Russell, G. (2013). *Patient-centred access to health care:* conceptualizing access at the interface of health systems and populations. International Journal for Equity and Health; 12 (18). Available at http://www.equityhealthj.com/content/12/1/18 (accessed 8/8/2015).

Loxton, D., Stewart Williams, J., Adamson, L. (2007). *Barriers to service delivery for young pregnant women and mothers*. National Youth Affairs Research Scheme. Available at

https://docs.education.gov.au/system/files/doc/other/barriers_to_service_delivery_for_young_pregnant_women_and_mothers.pdf (accessed 5/5/2017).

Macleod, C.I., & Tracey, T. (2010). A decade later: follow-up review of South African research on the consequences of and contributory factors in teen-aged pregnancy. South African Journal of Psychology, 40(1), 18-31.

Manerikar, V., Manerikar, S. (2014). *A note on exploratory research*. AWEshkar Journal, 17(1), 95-96.

Maness, S.B., Buhi, E.R., Daley, E.M., Baldwin, J.A., Kromrey, J.D. (2016). *Social determinants of health and adolescent pregnancy: an analysis from the national longitudinal study of adolescent to adult health.* Journal of Adolescent Health, 58, 636-643.

Mathews, C., Eggers, S.M., de Vries, P.J., Mason-Jones, A.J., Townsend, L., Aarø, L.E., De Vries, H. (2015). *Reaching the hard to reach: longitudinal investigation of adolescents' attendance at and after-school sexual and reproductive health programme in Western Cape, South Africa.* BMC Public Health, 15 (608). Available at https://bmcpublichealth.biomedcentral.com/articles/10.1186/s12889-015-1963-3 (accessed 8/8/2015).

Mayeye, F.B., Lewis, H.A., Oguntibeju, O.O. (2010). An Assessment of Adolescent Satisfaction with Reproductive Primary Healthcare Services in the Eastern Cape Province, South Africa. West Indian Medical Journal 2010, 59 (3), 274-279.

Mchunu, G., Peltzer, K., Tutshana, B., Seutlwadi, L. (2012). *Adolescent pregnancy and associated factors in South African youth*. African Health Sciences, 12 (4), 426-434. Available at https://www.ajol.info/index.php/ahs/article/view/85072 (accessed 11/6/2016).

Mfono, Z. (1998). *Teenage contraceptive needs in urban South Africa: a case study*. International Family Planning Perspectives, 24 (4), 180-183.

Morwe, K.G., Klu, E.K., Tugli, A.K. (2014). *Teenage pregnancy in South Africa: a challenge to democracy*. Journal of Social Sciences, 41(3), 481-485.

Mturi, A.J. (2015). Why young unmarried women bear children? A case study of North West province, South Africa. Collen Programme. Available at http://www.ageing.ox.ac.uk/download/155 (accessed 5/5/2017).

Mumah, J., Kabiru, CW., Izugbara, C., Mukiira, C. (2014). *Coping with unintended pregnancies: narratives from adolescents in Nairobi's slums*. African Population and Health Research Center. Available at http://www.popcouncil.org/uploads/pdfs/2014STEPUP_APHRC-ResearchReport.pdf (accessed 6/5/2017).

Mushwana L., Monareng L., Richter S., Muller H.(2015) Factors influencing the adolescent pregnancy rate in the Greater Giyani Municipality, Limpopo Province – South Africa. International Journal of African Nursing Sciences, 2, 10-18.

Mwaba, K. (2000). *Perceptions of teenage pregnancy among South African adolescents*. Health SA Gesondheid, 5 (3), 30-34.

Newton-Levinson, A., Leichleter, J.S., Chandra-Mouli, V. (2016). Sexually transmitted infection services for adolescents and youth in low-and middle-income countries: perceived and experienced barriers to accessing care. Journal of Adolescent Health, 59, 7-16.

Nisar, N., White, E. (2003). Factors affecting utilization of antenatal care among reproductive age group women (15-49 years) in an urban squatter settlement of Karachi. Journal of Pakistan Medical Association, 53(47). Available at http://jpma.org.pk/full_article_text.php?article_id=2065 (accessed 24/9/2015).

Nkani, N., Bhana, D. (2016). Sexual and reproductive well-being of teenage mothers in a South African township school. South African Journal of Education, 36 (2). Available at http://www.sajournalofeducation.co.za/index.php/saje/article/view/1181/625 (accessed 25/10/2016).

Nyarko, K., Adentwi, K.I., Asumeng, M., Ahulu, L.D. (2014). *Parental attitude towards sex education at the lower primary in Ghana*. International Journal of Elementary Education, 3 (2), 21-29. Available at http://www.sciencepublishinggroup.com/j/ijeedu (accessed 11/7/2015).

Onokerhoraye, A.G., Dudu, J.E. (2017). *Perception of adolescents on the attitudes of providers on their access and use of reproductive health services in Delta State, Nigeria.* Health, 9, 88-105. Available at http://www.scirp.org/journal/health. (accessed 8/2/2017).

Patton, M., Cochran, M. (2002). *A guide to using qualitative research methodology*. Medicins Sans Frontieres. Available online:

http://fieldresearch.msf.org/msf/bitstream/10144/84230/1/Qualitative%20research%20meth odology.pdf (Accessed 9/9/2015).

Pell, C., Meñaca, A., Were, F., Afrah, N.A., Chatio, S., Manda-Taylor, L., Hamel, M.J., Hodgson, A., Tagbor, H., Kalilani, L., Ouma, P., Pool, R. (2013). *Factors affecting antenatal care attendance: Results from qualitative studies in Ghana, Kenya, and Malawi*. PLoS ONE 8(1). Available at http://journals.plos.org/plosone/article?id=10.1371/journal.pone.0053747 (accessed

17/1/2017).

Phafoli, S.H., Van Aswegen, E.J., Alberts, U.U. (2007). *Variables influencing delay in antenatal clinic attendance among adolescents in Lesotho*. South African Family Practice, 49 (9). Available at www.safpj.co.za (accessed 10/7/2017).

Pope, C., Ziebland, S., Mays, N. (2000). *Qualitative research in health care. Analysing qualitative data*. British Medical Journal, 320, 114-116.

Raatikainen, K., Heiskanen, N., Verkasalo, P.K., Heinonen, S. (2005). *Good outcome of teenage pregnancies in high-quality maternity care*. European Journal of Public Health, 16 (2), 157-161.

Resty, N. (2011). Factors affecting the utilization of antenatal care services among adolescent pregnant mothers. Case study of Naguru Teenage Health Centre, Kampala, Uganda. International. Statistical Institute: Proc. 58th World Statistical Congress, 2011, Dublin (Session CPS001). Available at http://2011.isiproceedings.org/papers/950879.pdf (accessed 6/5/2017).

Richards, H.M., Schwartz, L.J. (2002). *Ethics in qualitative research: are there special issues for health services research?* Family Practice, 19 (2), 135-139.

Richter, MS., Mlambo, GT. (2005). *Perceptions of rural adolescents on teenage pregnancy*. Health SA Gesondheid, 10 (2), 61-69.

Richter, L.M., Norris, S.A., Ginsburg, C. (2006). The silent truth of teenage pregnancies-Birth to twenty cohort's next generation. South African Medical Journal, 96 (2), 122-124.

Ritchie, J., Lewis, J., Elam, G. (2003). *Chapter 4 – Designing and Selecting Samples in J. Ritchie and J. Lewis (eds). Qualitative Research Practice: A Guide for Social Science Students and Researchers.* London: Sage Publishers.

Robson, C. (2011). *Ch 17- The analysis and interpretation of qualitative data. Real World Research.* Chichester: Wiley, 465-494.

Rosenberg, M., Pettifor, A., Miller, W.C., Thirumurthy, H., Emch, M., Afolabi, S.A., Kahn, K., Collinson, M., Tollman, S. (2015). *Relationship between school dropout and teen pregnancy among rural South African young women*. International Journal of Epidemiology, 44 (3), 928-936.

Rukundo, G.Z., Abaasa, C., Natukunda, P.B., Ashabahebwa, B.H., Allain, D. (2015). *Antenatal services for pregnant adolescents in Mbarara Municipality, Southwestern Uganda: health workers and community leaders' views.* BMC Pregnancy and Childbirth, 15 (351). Available at https://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/s12884-015-0772-0 (accessed 17/1/2017).

Rutenberg, N., Kaufman, C.E., Macintyre, K., Brown, L., Karim, A. (2003). *Pregnant or positive: adolescent childbearing and HIV risk in KwaZulu Natal, South Africa*.

Reproductive Health Matters Journal, 11(22), 122-133.

Silal, S.P., Penn-Kekana, L., Harris, B., Birch, S., McIntyre, D. (2012). *Exploring inequalities in access to and use of maternal health services in South Africa*. BMC Health

Services Research 12 (120), 1-12. Available at http://www.biomedcentral.com/1472-6963/12/120 (accessed 6/5/2017).

Singh, S., Hamid, A. (2015). *Reflections of a group of South African teenage mothers: Sexual health implications.* Health Education Journal, 1-11.

Srivastava, A., Thomson, S.B. (2009). *Framework Analysis: A Qualitative Methodology for Applied Policy Research*. Journal of Administration and Governance, 4 (2), 72-79.

Statistics South Africa. (2016). *Statistical release: recorded live births 2013-2015*. Available online: www.statssa.gov.za/publications/P0305/P03052015.pdf (Accessed 20/11/2017).

Taffa, N., Obare, F. (2004). *Pregnancy and child health outcomes among adolescents in Ethiopia*. Ethiopian Journal of Health Development 18 (2), 90-95.

Tan, P.S., Tohid, H., Su, X.V., Tan, K.T.M., Azimah, M.N., Khairani, O. (2012). *A study on pregnant adolescents residing in a government home: common characteristics and their views on pregnancy*. Malaysian Family Physician, 7 (1), 11-15. Available at http://www.e-mfp.org/ (accessed on 6/5/2017)

Tegegn, A., Yazachew, M., Gelaw Y. (2008). *Reproductive health knowledge and attitude among adolescents: A community based study in Jimma Town, Ethiopia.* Ethiopian Journal of Health Development 22 (3), 143-151.

The World Bank. (2015.) *Data: Adolescent Fertility*. Available at www.data.worldbank.org/indicator/SP.ADO.TFRT/countries (Accessed 11/7/2015).

Thobejane, T.D. (2015). Factors contributing to teenage pregnancy in South Africa: The case of Matjitjileng Village. Journal of Sociology and Social Anthropology, 6(2), 273-277.

Tilahun, M., Mengistie, B., Egata, G., Reda, A.A. (2012). *Health workers' attitudes toward sexual and reproductive health services for unmarried adolescents in Ethiopia*.

Reproductive Health, 9 (19). Available at http://www.reproductive-health-journal.com/content/9/1/19 (accessed 7/5/2017)

Tsawe, M., Susuman, A. (2014). Determinants of access to and use of maternal health care services in the Eastern Cape, South Africa: a quantitative and qualitative investigation.

BMC Research Notes, 7 (723). Available at http://www.biomedcentral.com/1756-0500/7/723

UN. (2014). The millennium development goals report. Available at http://www.un.org/millenniumgoals/2014%20MDG%20report/MDG%202014%20English %20web.pdf (accessed 7/11/2015).

UNPFA. (2013). *Adolescent pregnancy: A review of the evidence*. New York. Available at http://www.unfpa.org/sites/default/files/pub-pdf/ADOLESCENT%20PREGNANCY_UNFPA.pdf (accessed 19/7/2015)

UNESCO. (2014). *Developing an education sector response to early and unintended pregnancy*. Available online: http://unesdoc.unesco.org/images/0023/002305/230510E.pdf (accessed 11/7/2015).

UNICEF. (2016). *UNICEF Data: Monitoring the situation of women and children*. Available online: https://data.unicef.org/topic/maternal-health/adolescent-health/ (accessed 15/8/2017).

USAID. (1996). *Conducting Key Informant Interviews*. Performance Monitoring and Evaluation Tips, Number 2. USAID Center for Development Information and Evaluation. Available at http://pdf.usaid.gov/pdf_docs/pnabs541.pdf (accessed 29/4/2016).

Van Zyl, L., van der Merwe, M., Chigeza, S. (2015). *Adolescents' lived experiences of their pregnancy and parenting in a semi-rural community in the Western Cape*. Social Work, 51 (2), 150-172.

Vundule, C., Maforah, F., Jewkes, R., Jordaan, E. (2001). *Risk Factors for Teenage Pregnancy Among Sexually Active Black Adolescents In Cape Town*. South African Medical Journal, 91 (1), 73-80.

Western Cape Government Provincial Treasury. (2013). *Regional Development Profile City of Cape Town*. Available at

www.westerncape.gov.za/assets/departments/treasury/Documents/Socio-economic-profiles/dcO_city_of_cape_town_ke_6-dec_2013.pdf (Accessed 19/7/2015)

WHO. (2008). *Adolescent pregnancy*. MPS Notes, 1 (1). Available online: www.who.int/maternal_child_adolescent/documents/mpsnnotes_2_lr.pdf (Accessed 20/11/2017.

WHO. (2014). Factsheet No 364. Available online: www.who.int/mediacentre/factsheet/fs364/en/ (Accessed 11/7/2015).

WHO. (2017). *Health topics: adolescent health*. Available online: www.who.int/topics/adolescent_health/en/ (Accessed 19/11/2017).

Wiemann, C.M., Rickert, V.I., Berenson, A.B., Volk, R.J. (2005). *Are pregnant adolescents stigmatized by pregnancy?* Journal of Adolescent Health, 36. Available at http://www.jahonline.org/article/S1054-139X(04)00456-2/abstract (accessed 27/10/2016).

Wood, K., Jewkes, R. (2006). *Blood blockages and scolding nurses: barriers to adolescent contraceptive use in South Africa*. Reproductive Health Matters Journal 14(27), 109-118.

Woog, V., Singh, S., Browne, A., Philbin, J. (2015). *Adolescent women's need for and use of sexual and reproductive health services in developing countries*. Guttmacher Institute. Available at www.guttmacher.org/pubs/Adolescent-SRHS-Need-Developing-Countries.pdf. (accessed 8/2/2017).

Worku, E.B., Woldesenbet, S.A. (2016). Factors that Influence Teenage Antenatal Care Utilization in John Taolo Gaetsewe (JTG) District of Northern Cape Province, South Africa: Underscoring the Need for Tackling Social Determinants of Health. International Journal of Maternal and Child Health and AIDS, 5, (2), 134-145.



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa

Tel: +27 21-959 2809 Fax: 27 21-959 2872

E-mail: soph-comm@uwc.ac.za

INFORMATION SHEET

Project Title: Knowledge, Attitudes, Beliefs and Perceptions Regarding Access to Maternal Health Care Amongst Pregnant Adolescents in the Mitchells Plain Sub-district

What is this study about?



This is a research project being conducted by Michelle Olivia Erasmus at the University of the Western Cape. We are inviting you to participate in this research project because you are a pregnant adolescent using the Mitchell's Plain Maternity and Obstetrics Unit (MOU) for your maternal health care needs. The purpose of this research project is to understand the perceptions and experiences of pregnant adolescents when accessing maternal health care services at Mitchells Plain MOU, in order to identify factors which may improve access to maternal health services by pregnant adolescents.

What will I be asked to do if I agree to participate?

You will be asked to give consent to participation. Once you have consented to participate, you will be asked to participate in an interview with the researcher. The interview will be audio recorded. The interview will take place at Mitchells Plain MOU, and should not take longer than one hour. This is a once off interview and you will not be required to do anything further regarding this research project on completion of the interview. You will be asked questions regarding your experiences of using Mitchells Plain MOU, your thoughts and opinions about what makes it easy and/or difficult to use Mitchells Plain MOU and how you perceive your interaction with the staff at Mitchells Plain MOU.

Would my participation in this study be kept confidential?

The researchers undertake to protect your identity and the nature of your contribution. To ensure your anonymity, your name will not be recorded on the interview guide. The answers you give will be given key codes which will give the researcher an indication of your key characteristics such as your age and stage of pregnancy, but only the researcher and the researcher's supervisor will have access to these codes. Furthermore, these codes will not allow anyone else to trace the information back to you personally. To ensure you confidentiality, all audio recordings and consent forms will be locked away, allowing no unauthorized access.

If we write a report or article about this research project, your identity will be protected.

In accordance with legal requirements and/or professional standards, we will disclose to the appropriate individuals and/or authorities information that comes to our attention concerning child abuse or neglect or potential harm to you or others. In this event, we will inform you that we have to break confidentiality to fulfil our legal responsibility to report to the designated authorities.

What are the risks of this research?

There may be some risks from participating in this research study. These risks include some discomfort or stress due to the sensitive nature of adolescent pregnancy.

All human interactions and talking about self or others carry some amount of risks. We will nevertheless minimize such risks and act promptly to assist you if you experience any discomfort, psychological or otherwise during the process of your participation in this study. Where necessary, an appropriate referral will be made to a suitable professional for further assistance or intervention.

What are the benefits of this research?

This research is not designed to help you personally, but the results may help the investigator learn more about the perspective and experiences of pregnant adolescents regarding access to maternal health care services. We hope that, in the future, other people might benefit from this study through improved understanding of the needs of pregnant adolescents.

Do I have to be in this research and may I stop participating at any time?

Your participation in this research is completely voluntary. You may choose not to take part at all. If you decide to participate in this research, you may stop participating at any time. If you decide not to participate in this study or if you stop participating at any time, you will not be penalized or lose any benefits to which you otherwise qualify.

What if I have questions?

This research is being conducted by Michelle Olivia Erasmus through the School of Public Health at the University of the Western Cape. If you have any questions about the research study itself, please contact Michelle Olivia Erasmus at:

Tel: 073 177 8758 or 021 391 0555

E-mail: erasmusmichelleolivia@gmail.com

Should you have any questions regarding this study and your rights as a research participant or if you wish to report any problems you have experienced related to the study, please contact:

Dr Lucia Knight
MPH Thesis co-ordinator
School of Public Health
University of the Western Cape
Private Bag X17
Bellville 7535
Iknight@uwc.ac.za



Prof José Frantz

Dean of the Faculty of Community and Health Sciences

University of the Western Cape

Private Bag X17

Bellville 7535

chs-deansoffice@uwc.ac.za

This research has been approved by the University of the Western Cape's Senate Research Committee. (REFERENCE NUMBER: BM/16/4/5)

BIOMEDICAL RESEARCH ETHICS ADMINISTRATION

Research Office

New Arts Building,

C-Block, Top Floor, Room 28

University of the Western Cape

Private Bag X17

Bellville 7535



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa *Tel: +27 21-959 2809 Fax: 27 21-959 2872*

E-mail: soph-comm@uwc.ac.za

INFORMATION SHEET

Project Title: Knowledge, Attitudes, Beliefs and Perceptions Regarding Access to Maternal Health Care Amongst Pregnant Adolescents in the Mitchells Plain Sub-district

What is this study about?

This is a research project being conducted by Michelle Olivia Erasmus at the University of the Western Cape. We are inviting your child to participate in this research project because she is a pregnant adolescent using the Mitchell's Plain Maternity and Obstetrics Unit (MOU) for your maternal health care needs. The purpose of this research project is to understand the perceptions and experiences of pregnant adolescents when accessing maternal health care services at Mitchells Plain MOU, in order to identify factors which may improve access to maternal health services by pregnant adolescents.

What will your child be asked to do if you and your child agree to participate?

You and your will be asked to give consent to participation. Once you and your child have consented to participate, your child will be asked to participate in an interview with the researcher. The interview will be audio recorded. The interview will take place at Mitchells Plain MOU, and should not take longer than one hour. This is a once off interview and you or your child will not be required to do anything further regarding this research project on completion of the interview. Your child will be asked questions regarding her experiences of using Mitchells Plain MOU, her thoughts and opinions about what makes it easy and/or difficult to use Mitchells Plain MOU and how she perceives her interaction with the staff at Mitchells Plain MOU.

Would my child's participation in this study be kept confidential?

The researchers undertake to protect your child's identity and the nature of her contribution. To ensure her anonymity, her name will not be recorded on the interview guide. The answers she gives will be given key codes which will give the researcher an indication of her key characteristics such as her age and stage of pregnancy, but only the researcher and the researcher's supervisor will have access to these codes. Furthermore, these codes will not allow anyone else to trace the information back to her personally. To ensure her confidentiality, all audio recordings and consent forms will be locked away, allowing no unauthorized access.

If we write a report or article about this research project, your child's identity will be protected.

In accordance with legal requirements and/or professional standards, we will disclose to the appropriate individuals and/or authorities information that comes to our attention concerning child abuse or neglect or potential harm to your child or others. In this event, we will inform you and/or your child that we have to break confidentiality to fulfil our legal responsibility to report to the designated authorities.

What are the risks of this research?

There may be some risks from participating in this research study. These risks include some discomfort or stress due to the sensitive nature of adolescent pregnancy.

All human interactions and talking about self or others carry some amount of risks. We will nevertheless minimize such risks and act promptly to assist your child if she experiences any discomfort, psychological or otherwise during the process of your participation in this study. Where necessary, an appropriate referral will be made to a suitable professional for further assistance or intervention.

What are the benefits of this research?

This research is not designed to help your child personally, but the results may help the investigator learn more about the perspective and experiences of pregnant adolescents regarding access to maternal health care services. We hope that, in the future, other people might benefit from this study through improved understanding of the needs of pregnant adolescents.

Do I have to be in this research and may I stop participating at any time?

Your child's participation in this research is completely voluntary. You may choose not to allow your child take part at all. If you decide that your child may participate in this research, you may withdraw the decision to allow your child to participate at any time. If you decide that your child may not participate in this study or if your child's participation at any time, your child will not be penalized or lose any benefits to which she otherwise qualifies.

What if I have questions?

This research is being conducted by Michelle Olivia Erasmus through the School of Public Health at the University of the Western Cape. If you have any questions about the research study itself, please contact Michelle Olivia Erasmus at:

Tel: 073 177 8758 or 021 391 0555

E-mail: erasmusmichelleolivia@gmail.com

Should you have any questions regarding this study and your rights as a research participant or if you wish to report any problems you have experienced related to the study, please contact:

Dr Lucia Knight
MPH Thesis co-ordinator
School of Public Health
University of the Western Cape
Private Bag X17
Bellville 7535
lknight@uwc.ac.za

Prof José Frantz

Dean of the Faculty of Community and Health Sciences

University of the Western Cape

Private Bag X17

Bellville 7535

This research has been approved by the University of the Western Cape's Senate Research Committee. (REFERENCE NUMBER: BM/16/4/5)

BIOMEDICAL RESEARCH ETHICS ADMINISTRATION

Research Office

New Arts Building,

C-Block, Top Floor, Room 28

chs-deansoffice@uwc.ac.za

University of the Western Cape

Private Bag X17

Bellville 7535



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa *Tel: +27 21-959 2809 Fax: 27 21-959 2872*

E-mail: soph-comm@uwc.ac.za

INFORMATION SHEET

Project Title: Knowledge, Attitudes, Beliefs and Perceptions Regarding Access to Maternal Health Care Amongst Pregnant Adolescents in the Mitchells Plain Sub-district

What is this study about?

This is a research project being conducted by Michelle Olivia Erasmus at the University of the Western Cape. We are inviting you to participate in this research project because you are a nurse working with pregnant adolescents at the Mitchell's Plain Maternity and Obstetrics Unit (MOU. The purpose of this research project is to understand the perceptions and experiences of pregnant adolescents when accessing maternal health care services at Mitchells Plain MOU, in order to identify factors which may improve access to maternal health services by pregnant adolescents. It would therefore be valuable to gain some insight from nursing staff working with pregnant adolescents at Mitchells Plain MOU.

What will I be asked to do if I agree to participate?

You will be asked to give consent to participation. Once you have consented to participate, you will be asked to participate in an interview with the researcher. The interview will be audio recorded. The interview will take place at Mitchells Plain MOU, and should not take longer than one hour. This is a once off interview and you will not be required to do anything further regarding this research project on completion of the interview. You will be asked questions regarding your experiences of being a caregiver to pregnant adolescents at Mitchells Plain MOU, your thoughts and opinions about what makes it easy and/or difficult to provide care to adolescents at Mitchells Plain MOU and how you perceive your interaction with pregnant adolescents at Mitchells Plain MOU.

Would my participation in this study be kept confidential?

The researchers undertake to protect your identity and the nature of your contribution. To

ensure your anonymity, your name will not be recorded on the interview guide. The answers you give will be given key codes which will give the researcher an indication of your key characteristics such as your age, but only the researcher and the researcher's supervisor will have access to these codes. Furthermore, these codes will not allow anyone else to trace the information back to you personally. To ensure you confidentiality, all audio recordings and consent forms will be locked away, allowing no unauthorized access.

If we write a report or article about this research project, your identity will be protected.

In accordance with legal requirements and/or professional standards, we will disclose to the appropriate individuals and/or authorities information that comes to our attention concerning child abuse or neglect or potential harm to you or others. In this event, we will inform you that we have to break confidentiality to fulfill our legal responsibility to report to the designated authorities.

What are the risks of this research?

There may be some risks from participating in this research study. These risks include some discomfort or stress due to the sensitive nature of adolescent pregnancy

All human interactions and talking about self or others carry some amount of risks. We will nevertheless minimize such risks and act promptly to assist you if you experience any discomfort, psychological or otherwise during the process of your participation in this study. Where necessary, an appropriate referral will be made to a suitable professional for further assistance or intervention.

What are the benefits of this research?

This research is not designed to help you personally, but the results may help the investigator learn more about the perspective and experiences of pregnant adolescents regarding access to maternal health care services. We hope that, in the future, other people might benefit from this study through improved understanding of the needs of pregnant adolescents.

Do I have to be in this research and may I stop participating at any time?

Your participation in this research is completely voluntary. You may choose not to take part at all. If you decide to participate in this research, you may stop participating at any time. If you decide not to participate in this study or if you stop participating at any time, you will not be penalized or lose any benefits to which you otherwise qualify.

What if I have questions?

This research is being conducted by Michelle Olivia Erasmus through the School of Public Health at the University of the Western Cape. If you have any questions about the research

study itself, please contact Michelle Olivia Erasmus at:

Tel: 073 177 8758 or 021 391 0555

E-mail: erasmusmichelleolivia@gmail.com

Should you have any questions regarding this study and your rights as a research participant or if you wish to report any problems you have experienced related to the study, please contact:

Dr Lucia Knight
MPH Thesis co-coordinator
School of Public Health
University of the Western Cape
Private Bag X17
Bellville 7535
lknight@uwc.ac.za

Prof José Frantz

Dean of the Faculty of Community and Health Sciences

University of the Western Cape

Private Bag X17

Bellville 7535

WESTERN CAPE

chs-deansoffice@uwc.ac.za

This research has been approved by the University of the Western Cape's Senate Research Committee. (REFERENCE NUMBER: BM/16/4/5)

BIOMEDICAL RESEARCH ETHICS ADMINISTRATION

Research Office
New Arts Building,
C-Block, Top Floor, Room 28
University of the Western Cape
Private Bag X17
Bellville 7535

RECORD OF INFORMED CONSENT TO CONDUCT AN INTERVIEW

CONSENT FORM

Title of Research Project: Knowledge, Attitudes, Beliefs and Perceptions Regarding Access to Maternal Health Care amongst Pregnant Adolescents in the Mitchells Plain Sub-district

The study has been described to me in language that I understand. My questions about the study have been answered. I understand what my participation will involve and I agree to participate of my own choice and free will. I understand that my identity will not be disclosed to anyone. I understand that I may withdraw from the study at any time without giving a reason and without fear of negative consequences or loss of benefits.

WESTERN CAPE

Participant's name	
Participant's signature	
Date	
BIOMEDICAL RESEARCH ETHICS ADM	MINISTRATION
Research Office	
New Arts Building,	<u> </u>
C. Black Top Floor Poom 29	UNIVERSITY of the

Private Bag X17

C-Block, Top Floor, Room 28

University of the Western Cape

Bellville 7535

Bellville 7535

RECORD OF INFORMED CONSENT TO CONDUCT AN INTERVIEW

PARENTAL PERMISSION FORM

Title of Research Project: Knowledge, Attitudes, Beliefs and Perceptions Regarding Access to Maternal Health Care amongst Pregnant Adolescents in the Mitchells Plain Sub-district

The study has been described to me in language that I understand. My questions about the study have been answered. I understand what my child's participation will involve and I agree to allow my child to participate of my own choice and free will. I understand that my child's identity will not be disclosed to anyone. I understand that I may withdraw my child from the study at any time without giving a reason and without fear of negative consequences or loss of benefits to me or my child

Parent/guardian's name	•••••
Parent/guardian's signature	
Date	
	11-11-11-11-11
BIOMEDICAL RESEARCH ETHICS ADM	NINISTRATION
Research Office	
New Arts Building,	·
C-Block, Top Floor, Room 28	UNIVERSITY of the
University of the Western Cape	WESTERN CAPE
Private Bag X17	

RECORD OF INFORMED CONSENT TO CONDUCT AN INTERVIEW

ASSENT FORM

Title of Research Project: Knowledge, Attitudes, Beliefs and Perceptions Regarding Access to Maternal Health Care amongst Pregnant Adolescents in the Mitchells Plain Sub-district

The study has been described to me in language that I understand. My questions about the study have been answered. I understand what my participation will involve and I agree to participate of my own choice and free will. I understand that my identity will not be revealed to anyone. I understand that I may decide to no longer participate in the study at any time without giving a reason and without fear of negative consequences or loss of benefits to me.

UNIVERSITY of the

WESTERN CAPE

Participant's name	
Participant's signature	······································
Date	

BIOMEDICAL RESEARCH ETHICS ADMINISTRATION

Research Office

New Arts Building,

C-Block, Top Floor, Room 28

University of the Western Cape

Private Bag X17

Bellville 7535

Interview guide- Adolescent

Introduce topic

The purpose of this research is to explore knowledge, attitudes beliefs and perceptions of pregnant adolescents regarding access to maternal health care services. The reason for this is that adolescents/young people often come to the MOU very late in pregnancy, so I want to find out reasons for this, and what makes it easier or more difficult for young people to come to the MOU. I hope that this research will then help to identify ways to improve services for young people.

• Gain demographic information (age, area of residence, school/employment, family background)

How old are you?

Where do you live?

Do you attend school/work/

If yes, what grade? If no, what grade did you finish?

Prompt for those who left school... Did you leave school due to pregnancy?

Who do you currently live with? WESTERN CAPE

• Can you tell me about your experience of finding out that you were pregnant?

Probe: How did that make you feel? Can you tell me about whether anyone helped you with this and how?

• Can you tell me about why you came to the MOU?

Probe: Can you tell me about whether anyone helped you decide to come here?

Probe: Can you tell me about anything that happened that made you decide to come here?

• Can you tell me about what it was like coming here to the MOU for the first time?

Probe: How did you feel about coming to the MOU?

Can you tell me about any good experiences you had at the MOU? How did that make you feel?

Can you tell me about any bad experiences you had at the MOU? How did that make you feel?

- Can you explain if there is anything that made it easier for you to come to the MOU?
- Can you explain if there is anything that made it difficult for you to come to the MOU?



Interview Guide: Nursing staff

- (Get demographic information: Age, years of experience)
- How do you feel about the high rate of teenage pregnancy in this community?

Probe: Can you tell me about things you think contribute to high rates of teenage pregnancy?

- Can you tell about why access to the MOU is important for pregnant teenagers?
- How do you feel about working with pregnant adolescents?

Probe: Can you tell me about any positive experiences/incidents when working with pregnant adolescents? How did that make you feel?

Can you tell me about any negative experiences/incidents when working with pregnant adolescents? How did that make you feel?

How would you describe your interaction with pregnant adolescents at the facility?

Probe: Can you explain how you react to pregnant teenagers who miss appointments/don't comply with medical advice?

Probe: Can you explain how you react to pregnant teenagers who seem nervous/anxious when they come to the MOU?

Can you tell me about reasons why adolescents make late bookings?

Probe: Can you explain if there is anything at the MOU that contributes to teenagers making late bookings?

• Can you explain what you think would help adolescents' access maternal health care?

Probe: Can you tell me about what you think staff can do differently to assist pregnant teenagers at the MOU?

Probe: Can you tell me about anything at the MOU you would like to see change to help you work more effectively with pregnant teenagers?